

UNIVERSIDAD NACIONAL AUTÓNOMA DE MÉXICO POSGRADO EN CIENCIAS BIOLÓGICAS

INSTITUTO DE BIOLOGÍA BIOLOGÍA EVOLUTIVA

(PROYECTO)

DEMOGRAFÍA HISTÓRICA DE POLIOPTILA CALIFORNICA (AVES: POLIOPTILIDAE) UN AVE

ENDÉMICA DE BAJA CALIFORNIA

TESIS

(POR ARTÍCULO CIENTÍFICO)

SYNCHRONOUS DEMOGRAPHIC AND DISTRIBUTION SHIFTS THROUGH TIME IN AN ENDANGERED

DESERT SPECIALIST

QUE PARA OPTAR POR EL GRADO DE:

MAESTRA EN CIENCIAS BIOLÓGICAS

PRESENTA:

CAMILA MARTÍNEZ AVILA

TUTOR PRINCIPAL DE TESIS: DR. HERNÁN VÁZQUEZ MIRANDA INSTITUTO DE BIOLOGÍA, UNAM COMITÉ TUTOR: DRA. ELLA GLORIA VÁZQUEZ DOMÍNGUEZ INSTITUTO DE ECOLOGÍA, UNAM COMITÉ TUTOR: DRA. CAROLINA GRANADOS MENDOZA INSTITUTO DE BIOLOGÍA, UNAM

CIUDAD UNIVERSITARIA, CD. MX., OCTUBRE 2023.



Universidad Nacional Autónoma de México



UNAM – Dirección General de Bibliotecas Tesis Digitales Restricciones de uso

DERECHOS RESERVADOS © PROHIBIDA SU REPRODUCCIÓN TOTAL O PARCIAL

Todo el material contenido en esta tesis esta protegido por la Ley Federal del Derecho de Autor (LFDA) de los Estados Unidos Mexicanos (México).

El uso de imágenes, fragmentos de videos, y demás material que sea objeto de protección de los derechos de autor, será exclusivamente para fines educativos e informativos y deberá citar la fuente donde la obtuvo mencionando el autor o autores. Cualquier uso distinto como el lucro, reproducción, edición o modificación, será perseguido y sancionado por el respectivo titular de los Derechos de Autor.



UNIVERSIDAD NACIONAL AUTÓNOMA DE MÉXICO POSGRADO EN CIENCIAS BIOLÓGICAS

INSTITUTO DE BIOLOGÍA BIOLOGÍA EVOLUTIVA

(PROYECTO)

DEMOGRAFÍA HISTÓRICA DE POLIOPTILA CALIFORNICA (AVES: POLIOPTILIDAE) UN AVE

ENDÉMICA DE BAJA CALIFORNIA

TESIS

(POR ARTÍCULO CIENTÍFICO)

SYNCHRONOUS DEMOGRAPHIC AND DISTRIBUTION SHIFTS THROUGH TIME IN AN ENDANGERED

DESERT SPECIALIST

QUE PARA OPTAR POR EL GRADO DE:

MAESTRA EN CIENCIAS BIOLÓGICAS

PRESENTA:

CAMILA MARTÍNEZ AVILA

TUTOR PRINCIPAL DE TESIS: DR. HERNÁN VÁZQUEZ MIRANDA INSTITUTO DE BIOLOGÍA, UNAM COMITÉ TUTOR: DRA. ELLA GLORIA VÁZQUEZ DOMÍNGUEZ INSTITUTO DE ECOLOGÍA, UNAM COMITÉ TUTOR: DRA. CAROLINA GRANADOS MENDOZA INSTITUTO DE BIOLOGÍA, UNAM

CIUDAD UNIVERSITARIA, CD. MX., OCTUBRE 2023.





COORDINACIÓN GENERAL DE ESTUDIOS DE POSGRADO COORDINACIÓN DEL POSGRADO EN CIENCIAS BIOLÓGICAS INSTITUTO DE BIOLOGÍA OFICIO: CGEP/CPCB/IB/0778/2023 ASUNTO: Oficio de Jurado

M. en C Ivonne Ramírez Wence Directora General de Administración Escolar, UNAM P r e s e n t e

Me permito informar a usted que en la reunión ordinaria del Comité Académico del Posgrado en Ciencias Biológicas, celebrada el día **31 de julio de 2023** se aprobó el siguiente jurado para el examen de grado de **MAESTRA EN CIENCIAS BIOLÓGICAS** en el campo de conocimiento de **BIOLOGÍA EVOLUTIVA** de la alumna **MARTÍNEZ AVILA CAMILA** con número de cuenta **314280350** por la modalidad de graduación de tesis por artículo científico titulado: "Synchronous demographic and distribution shifts through time in an endangered desert specialist", que es producto del proyecto realizado en la maestría que lleva por título: "Demografía histórica de *Polioptila californica* (Aves: Polioptilidae) un ave endémica de Baja California", ambos realizados bajo la dirección del DR. HERNÁN VÁZQUEZ MIRANDA, quedando integrado de la siguiente manera:

Presidente:	DR. ADOLFO GERARDO NAVARRO SIGÜENZA
Vocal:	DR. LUIS ANTONIO SÁNCHEZ GONZÁLEZ
Vocal:	DR. SANTIAGO ALEJANDRO RAMÍREZ BARAHONA
Vocal:	DRA. CLAUDIA PATRICIA ORNELAS GARCÍA
Secretaria:	DRA. CAROLINA GRANADOS MENDOZA

Sin otro particular, me es grato enviarle un cordial saludo.

A T E N T A M E N T E "POR MI RAZA HABLARÁ EL ESPÍRITU" Ciudad Universitaria, Cd. Mx., a 27 de septiembre de 2023

COORDINADOR DEL PROGRAMA

DR. ADOLFO GERARDO NÁVARRO SIGÜENZA



c. c. p. Expediente del alumno

AGNS/RGA/EARR/rga

COORDINACIÓN DEL POSGRADO EN CIENCIAS BIOLÓGICAS Unidad de Posgrado, Edificio D, 1º Piso. Circuito de Posgrados, Ciudad Universitaria Alcaldía Coyoacán. C. P. 04510 CDMX Tel. (+5255)5623 7002 http://pcbiol.posgrado.unam.mx/

Agradecimientos Institucionales

Agradezco a todas las instituciones y personas que me apoyaron en cada una de las etapas de este proyecto. Primeramente, al Posgrado en Ciencias Biológicas de la Universidad Nacional Autónoma de México quien me brindó las herramientas básicas necesarias para desarrollarme y llevar a cabo esta investigación.

Agradezco al Consejo Nacional de Humanidades, Ciencias y Tecnología (CONAHCYT) por la beca de maestría otorgada mediante el programa de Becas Tradicionales (Número de CVU 1101167). Sin este apoyo económico no habría sido posible enfocarme a mis estudios y obtener el grado de maestría. Agradezco también al Programa de Apoyo a Proyectos de Investigación e Innovación Tecnológica DGAPA-PAPIIT (IA204220 y IA205422) por el financiamiento que hizo posible este proyecto. Asimismo, agradezco a la UNAM por la beca PAPIIT (Clave de becario B231330) que me fue asignada como apoyo durante la etapa de titulación.

Agradezco de manera especial a mi tutor, el Dr. Hernán Vázquez Miranda, quien me recibió en su laboratorio y me permitió comenzar esta etapa clave en mi formación. Igualmente agradezco a los miembros de mi comité tutoral, la Dra. Ella Gloria Vázquez Domínguez y la Dra. Carolina Granados Mendoza por los comentarios y acompañamiento durante la realización de este proyecto.

Agradecimientos Personales

A Hernán, por abrirme las puertas de su laboratorio, por introducirme al mundo de la genómica y al de las aves. Por enseñarme todo lo que sé de la academia y por impulsarme a hacer más.

A las doctoras Carolina y Ella, por los comentarios y correcciones que robustecieron este trabajo y, sobre todo, por su orientación y ayuda a lo largo de la maestría, sobre todo

A los miembros de mi jurado, la Dra. Claudia Patricia Ornelas García, el Dr. Adolfo Gerardo Navarro Sigüenza, el Dr. Luis Antonio Sánchez González y el Dr. Santiago Alejandro Ramírez Barahona, por sus revisiones y comentarios.

A Sali, por ser mi cómplice desde siempre y mi inspiración para seguir empujando los límites. Gracias por tanto que me has enseñado, por las pláticas, las risas y el apoyo.

A Tomás, por ser mi compañía desde el día uno, por el apoyo en forma de porras, tardes de trabajo en el café y cenas. Gracias por el amor y el refugio que me has dado.

A Brenda, por todo lo compartido a través de los años, por la energía y fuerza que me inyectas y por la ayuda para salir de las temporadas de marchite.

A Viole y David, por ser mi guía en el posgrado, por apoyarme en todo momento, por las enseñanzas y por su invaluable amistad.

A todos los amigos, familiares y profesores que han sido parte de este camino y me ayudaron a llegar a este punto.

Y, por último, quiero dedicarle este trabajo a Blanca y a Manolo, mis papás. Quienes desde niña me han impulsado a cumplir todos mis sueños y me brindaron la oportunidad de convertirme en la persona que soy ahora. Sin su arduo trabajo, amor incondicional y confianza jamás lo hubiera logrado.

Índice

Resumen <u>1</u>
Abstract
Introducción
Artículo científico
Title and authors <u>7</u>
Abstract
Significance statement
Introduction <u>9</u>
Results
Discussion
Materials and methods
Data accessibility
Funding
Acknowledgements
References
Discusión y conclusiones
Referencias bibliográficas
Apéndices

Resumen

Las oscilaciones climáticas que ocurrieron durante el periodo Cuaternario modificaron la distribución de los ecosistemas alrededor del mundo, especialmente en las zonas cercanas a los polos. Estos cambios ambientales afectaron a las especies que habitaban estas regiones, de modo que dichas especies presentan firmas genómicas y demográficas particulares que nos brindan información sobre los fenómenos y procesos que han experimentado.

El objetivo de este estudio fue evaluar algunos de los procesos históricos ambientales que moldearon genómica y espacialmente a la perlita de California, *Polioptila california*, un ave pequeña que habita principalmente en los matorrales costeros de las zonas áridas de la península de Baja California. Para evaluar de manera más integral la historia evolutiva de la especie empleamos dos herramientas analíticas distintas: los modelos de demografía histórica y los modelos de distribución de especies.

Los resultados mostraron que las poblaciones de la perlita de California pasaron por un severo cuello de botella al finalizar el Último Máximo Glacial (UMG), cuando el hábitat disponible para la especie estaba restringido a la porción más sureña de la península. Este declive poblacional produjo una pérdida significativa de diversidad genética de la cual *P. california* no ha logrado recuperarse. No encontramos evidencia de estructura genética en la perlita de California, de manera que nuestros resultados no apoyan ninguna de las hipótesis de subespecies planteadas previamente por otros autores.

A pesar de que es bien reconocido el rol fundamental de los factores ambientales en la determinación de las tendencias evolutivas de las poblaciones naturales, en algunos casos aún no sabemos de manera precisa cuáles son los procesos que incidieron en las poblaciones ni cómo influyen en propiedades biológicas como el tamaño o la estructura poblacional. Aquí encontramos que los cambios en la distribución geográfica probablemente jugaron un papel clave en los procesos demográficos de un ave peninsular. Este estudio presenta un ejemplo de cómo variaciones ambientales naturales de gran magnitud han influido en la evolución de las especies.

Abstract

The climatic oscillations that took place during the Quaternary period modified the ecosystems' distributions around the world, especially near the poles. These environmental changes affected the species inhabiting these regions, so that those species have particular genomic and demographic signatures that give us information about the phenomena and processes that these species have experienced.

The objective of this study was to evaluate some of the historical environmental processes that modelled genomically and spatially the California gnatcatcher, *Polioptila californica*, a small avian species that lives primarily on the coastal scrubs from the arid lands on the Baja California Peninsula. To assess more integrally the species' evolutionary history, we used two different analytical tools: the historical demography models and the species' distribution models.

The results showed that the California gnatcatcher's populations went through a severe bottleneck after the end of the Last Glacial Maximum (LGM), when the available habitat for the species was restricted to the southern part of the peninsula. This population decline produced a significant loss of genetic diversity from where *P. californica* has not been able to recover yet. We did not find evidence of genetic structure in the California gnatcatcher; thus, our results do not support any of the subspecies' hypotheses proposed by other authors.

Even though the fundamental role of the environmental factors on the determination of populations' evolutionary tendencies is well known, in some cases we still do not know what precise processes affected the populations nor how these mechanisms influence on biological properties such as population size or structure. Here we found that the geographic distributional changes likely played a central role in the demographic processes of a peninsular bird. This study presents an example of how major environmental changes have influenced species' evolution.

Introducción

Elucidar los procesos que promueven la divergencia de linajes es una pregunta central en biología evolutiva. Múltiples hipótesis se han planteado para explicar el origen de nuevas especies a partir de las poblaciones preexistentes. Probablemente el mecanismo más frecuentemente propuesto en grupos animales es la especiación alopátrida, la cual plantea que el surgimiento de barreras físicas en la distribución geográfica de una especie puede promover la separación de la población original en dos o más grupos de individuos. Eventualmente, los grupos derivados de la separación comienzan a divergir por acción de la deriva génica o las presiones de selección (Hoskin et al., 2005). Comúnmente, se piensa en barreras físicas como ríos o montañas que evitan la dispersión (i.e., flujo génico); sin embargo, la baja disponibilidad de sitios ecológicamente óptimos para las especies también puede representar una barrera al flujo génico. De este modo, cuando las especies no encuentran las condiciones para sobrevivir, los individuos pueden quedar geográficamente aislados de otros. Así, las barreras al flujo génico dadas por cambios en la distribución de las especies y pérdida de la conectividad del paisaje pueden derivar en la divergencia de linajes (Klinga et al., 2019; Pyron y Burbrink, 2010).

Por lo anterior, regiones geográficas que recientemente han sufrido eventos tectónicos u orográficos son buenos modelos para el estudio de la divergencia de linajes y la especiación. Un ejemplo de ello es la península de Baja California, la cual ha sido llamada un laboratorio natural (Álvarez-Castañeda y Murphy, 2014; Brown, 1987) debido a su historia geológica (Sedlock, 2003) y su alta diversidad de biomas, lo que ha provocado que las poblaciones que allí habitan tengan historias biogeográficas particulares y contrastantes entre sí. Por ejemplo, Riddle et al. (2000) revisó los análisis filogeográficos previos de 12 especies que se distribuyen en la península y encontró cuatro patrones geográficos diferentes que reflejaban la historia biogeográfica de los grupos genéticos en estas especies. Dichos autores hacen hincapié en la independencia de la biota de la península y la del desierto sonorense. También existen estudios que proponen que un único evento tectónico u orogénico no sería suficiente para crear los complicados patrones filogeográficos que encontramos en las especies peninsulares. Tal es el caso de un estudio hecho con seis mamíferos y seis reptiles en donde se sugirió que los efectos combinados de las incursiones marinas y el vulcanismo son los que explican de mejor manera la complejidad de estas historias (Leaché et al., 2007).

Se ha reportado que distintos eventos geológicos y climáticos que ocurrieron en la península, o cerca de la región peninsular, han inducido procesos de especiación, endemismo y fragmentación de las áreas de distribución (Klimova et al., 2017). Tres ejemplos de estos grandes eventos geológicos son la formación de un canal marino en la porción media de la península durante el Pleistoceno, la transgresión hacia el norte del mar de Cortés durante el Plioceno tardío, y la formación de un canal marino en el Plioceno que atravesaba el istmo de La Paz (Riddle et al., 2000). Es importante recalcar que la península está situada sobre múltiples placas tectónicas que se comportan de manera diferente. La parte occidental se encuentra sobre una falla transformante, mientras que la porción oriental concuerda con un límite divergente. Recientemente, se definieron ocho unidades biogeográficas dentro de la península de Baja California con base en la geología y las comunidades biológicas que habitan en el área, cada una representando una biota distinta. Las ocho unidades son: Desierto Central, Costa Central del Golfo, Isla de Guadalupe, Desierto del Vizcaíno, Sierra de la Giganta, Llanuras de Magdalena, Cabo y Sierra de la Laguna (Morrone, 2021).

Además de las particularidades de la península de Baja California, los eventos de magnitud global que ocurrieron durante las glaciaciones, como los cambios en el nivel del mar, la temperatura global y la precipitación (Hewitt, 2004), deben ser tomados en cuenta cuando buscamos inferir la historia evolutiva de una especie. El periodo Cuaternario (2.5 millones de años-actualidad) se ha caracterizado por presentar numerosos periodos glaciales e interglaciales (Bennett, 1990; Mantooth et al., 2013). Sin embargo, en los últimos 245,000 años han ocurrido múltiples glaciaciones de magnitudes variantes. La última glaciación comenzó hace aproximadamente 110,000 años, pero no fue hasta hace 20,000 años cuando la glaciación alcanzó su pico, iniciando así uno de los eventos climáticos más importantes del Pleistoceno: el Último Máximo Glacial, en el cual los glaciales de todo el mundo alcanzaron su tamaño máximo. Este evento tuvo mayor impacto en las regiones cerca de los polos donde la radiación solar es menor, lo que provocó que las zonas adyacentes sufrieran bajas considerables en la temperatura ambiental, favoreciendo la expansión de los glaciales. De este modo, las zonas templadas del hemisferio Norte fueron empujadas hacia el Ecuador reduciendo drásticamente el área de los ecosistemas templados (Williams et al., 2004). Un caso especial fue el de los desiertos del oeste de Norteamérica. La extensión de estos ecosistemas se redujo significativamente durante el periodo glacial del Pleistoceno, y no fue sino hasta que finalizó el periodo glacial (hace aproximadamente 10,000 años) que los desiertos recuperaron territorio, alcanzando su distribución máxima durante el Óptimo Climático del Holoceno (hace aproximadamente 8,000 años) (Mantooth et al., 2013; Riddle y Hafner, 2006).

Así como los biomas se redujeron, durante el UMG muchas poblaciones de regiones tropicales y áridas sufrieron cuellos de botella y fragmentación poblacional debido a la contracción de sus hábitats (Hewitt, 2000, 2004; Ony et al., 2021). Contrario a esto, con el establecimiento del periodo interglaciar y la regresión de los glaciales, las poblaciones del Hemisferio Norte pasaron por una explosión demográfica y una expansión en la distribución geográfica hacia el norte del continente (Nason et al., 2002). Por eso se ha propuesto que particularmente las especies de zonas áridas de

Norteamérica presentan firmas genéticas de divergencia y posterior diversificación que concuerdan con los cambios climáticos y geográficos de la región durante el Cuaternario. Específicamente, en los genomas de especies asociadas a ambientes áridos existe evidencia de aislamiento poblacional. Esto se ha asociado con grupos de individuos que quedaron aislados en refugios pleistocénicos y, tras el fin del UMG, cuando las condiciones climáticas fueron más adecuadas y se recuperó la vegetación, recobraron la conectividad y el flujo génico con el resto de la población (Graham et al., 2020; Sánchez-del Pino et al., 2020; Wilson y Pitts, 2012).

El uso combinado de herramientas genómicas y de modelos de distribución de especies (SDM, por sus siglas en inglés) permiten evaluar preguntas biológicas desde distintas perspectivas, de manera que, si los resultados de ambos enfoques concuerdan, las conclusiones que obtenemos adquieren mayor robustez (Aguirre-Liguori et al., 2021). Por un lado, los métodos de estimación de tamaño efectivo a partir de datos genómicos han ganado mucha popularidad en distintas áreas como la biología evolutiva, la ecología o la genómica de la conservación. Saber cómo han reaccionado las poblaciones en el pasado ante eventos específicos puede brindarnos información para inferir cómo podrían actuar en el futuro bajo escenarios particulares como la pérdida o fragmentación del hábitat, el calentamiento global y la disponibilidad de agua (Prates et al., 2016). Por otro lado, los SDM nos permiten visualizar los cambios que han ocurrido en el espacio geográfico con respecto a las condiciones climáticas y los recursos ambientales necesarios para la supervivencia de las especies, siempre asociado con el nicho ecológico particular de cada especie (Miller, 2010). De este modo, estudios que evalúan a la par los cambios distribucionales y la demografía poblacional pueden ayudarnos a comprender de mejor manera cómo los cambios climáticos de gran escala han moldeado las poblaciones actuales.

En este estudio buscamos evaluar si los cambios climáticos que ocurrieron durante el UMG redujeron la extensión de las zonas áridas de la península de Baja California y si esto provocó que las especies que habitan estos desiertos pasaran por cuellos de botella poblacionales asociados con la reducción de su hábitat. Para ello usamos como especie modelo a la perlita de California, *Polioptila californica*, una pequeña ave paseriforme endémica de la península de Baja California. Esta ave se encuentra asociada particularmente con regiones áridas y semiáridas desde el sur de California, EE.UU., hasta la punta sur de Baja California Sur. Estudios previos han encontrado evidencia limitada de una reciente expansión poblacional que concuerda temporalmente con la conclusión del UMG (Zink et al., 2013), así como falta de estructura genética y aislamiento geográfico a lo largo de la península (Vázquez-Miranda et al., 2022; Zink et al., 2016). Lo anterior sugiere que la especie no ha sufrido fragmentación, que el tiempo de aislamiento no fue suficiente para inducir divergencia, o

bien, que posterior a la divergencia el flujo génico fue tal que permitió la homogenización genética de las poblaciones.

Actualmente existe un debate sobre la clasificación infraespecífica de *Polioptila californica*, dado que se han planteado múltiples hipótesis de subespecies (algunas de ellas resumidas por Zink et al. 2000). Particularmente, se ha propuesto que las poblaciones que habitan en los matorrales costeros de EE.UU. pertenecen a una entidad taxonómica distinta clasificada como la subespecie *P. californica californica* (McCormack y Maley, 2015). Esta hipótesis está basada en la cantidad de color blanco en las plumas de la cola, vocalizaciones ligeramente más agudas y diferencias en la preferencia de hábitat con respecto a las aves del resto de la península. Esta subespecie se encuentra listada en el Acta de Especies (ESA, por sus siglas en inglés) de EE.UU. (ESA; 16 USC 1531 et seq; U.S. Fish and Wildlife Services 1995) y ha sido usada como especie bandera para la conservación de los matorrales de costa. Los estudios mencionados anteriormente fueron realizados con unos pocos genes mitocondriales y nucleares, por lo que realizar un estudio usando un mayor número de genes nucleares no ligados (independientes) puede permitir inferir con mayor solidez la historia evolutiva de la especie. Todo lo anterior hace a la perlita de California un modelo ideal para estudiar el efecto de los cambios climáticos en la estructura genética, la demografía y la distribución geográfica de las especies peninsulares.

1	TITL	Æ

2 Synchronous demographic and distribution shifts through time in an endangered desert specialist.3

4	AUTHORS AND	AFFILIATIONS
---	--------------------	---------------------

- 5 Camila Martínez-Avila^{1,2}, Carolina Granados Mendoza³, Ella Vázquez Domínguez⁴, and Hernán
- 6 Vázquez-Miranda^{1,*}.
- 7 ¹Colección Nacional de Aves, Departamento de Zoología, Instituto de Biología, Universidad
- 8 Nacional Autónoma de México, CDMX 04510.
- 9 ²Posgrado en Ciencias Biológicas, Universidad Nacional Autónoma de México, CDMX 04510.
- 10 ³Departamento de Botánica, Instituto de Biología, Universidad Nacional Autónoma de México,
- 11 CDMX 04510.
- ⁴Laboratorio de Genética y Ecología, Departamento de Ecología de la Biodiversidad, Instituto de

13 Ecología, Universidad Nacional Autónoma de México, 04510.

14 *Corresponding author: E-mail: hernan@ib.unam.mx.

- 15
- 16
- 17
- 18
- 19
- 20
- 21
- 22
- 23
- 24
- 27
- 25
- 26

27 ABSTRACT

The climatic oscillations that took place during the Quaternary period altered the ecosystems 28 29 distributions around the world. Different biomes contracted and expanded as temperature and 30 precipitation fluctuated over time yielding characteristic demographic signatures in the genome of species associated to changing habitats. Here, using coalescent-based historical demography and 31 32 species distribution models, we assessed the demographic processes experienced by an arid-land 33 bird species from the Baja California Peninsula. We evaluated how the climatic and environmental 34 changes that occurred during and after the last glaciation affected the population of the California gnatcatcher (*Polioptila californica*). We found that the gnatcatcher's population went through an 35 36 extreme genetic bottleneck around the end of the Last Glacial Maximum, when its suitable habitat 37 was almost entirely restricted to the south of the Peninsula. Population decline led to a significant loss of genetic diversity from which the species has not vet recovered. We did not find evidence for 38 subpopulation structure. Our study is an example of the effects of historic climatic changes on 39 40 species with arid-like habitat preferences and mid-low dispersal ability.

Keywords: distributional shifts, genomics, historical demography, Pleistocene, *Polioptila californica*.

43

44 SIGNIFICANCE STATEMENT

Our study formally tests the role of environmental factors in modelling the evolution of biological populations. Specifically, we test which genomic and demographic processes affect biological properties like demography, genetic levels and subpopulation structure. Here we provide evidence in favor that changes in the geographic range play a major role in the population size of a peninsular avian species adapted to arid environments. Our study highlights the profound impact that drastic historical environmental variations can have on a species' population genetic diversity.

51

52 **INTRODUCTION**

Determining the processes promoting lineage divergence is a central question in 53 54 evolutionary biology. Several hypotheses have been proposed to explain the origin of new species 55 from preexisting populations. Probably the most commonly proposed mechanism in animals is the allopatric speciation which states that the emergence of geographic barriers in an otherwise 56 57 continuous distribution of a species splits the original population in two or more groups. As time 58 goes by, these populations start diverging due to the action of genetic drift or selective pressures 59 (Hoskin et al., 2005). Normally, we think of these barriers as rivers or mountains that restrain 60 dispersal and gene flow of individuals from both sides of the barrier. Nevertheless, the absence of 61 available and suitable habitat can also act as a barrier. Consequently, species that do not have the 62 conditions to thrive in these unsuitable places, can become isolated in geographically distinct spaces. Therefore, changes in species distribution and landscape connectivity constitute barriers to 63 gene flow that can drive divergence (Klinga et al., 2019; Pyron & Burbrink, 2010). 64

Geographic regions that have recently experienced tectonic or orographic events are 65 66 excellent model systems for studying speciation. This is the case of the Baja California Peninsula, which has been called a natural laboratory (Álvarez-Castañeda & Murphy, 2014; Brown, 1987) due 67 to its intricate geologic history (Sedlock, 2003) and high biome diversity, yielding populations with 68 69 rather particular and varied biogeographic histories. For example, Riddle et al. (2000) evaluated phylogeographic analyses from 12 species distributed on the Baja California peninsula and found 70 71 four different geographic patterns related with the location of genetically distinct groups. In this 72 study, the authors highlight the independency of the peninsular biotas from those of the Sonoran 73 Desert. There are also studies proposing that a single event is not sufficient to create the complex 74 phylogeographical patterns of the peninsular biota. This is the case of a study made with six mammals and six reptiles suggest that the combined effect of water incursions and volcanism may 75 76 be a better explanation for the complexity of these patterns (Leaché et al., 2007).

77 Furthermore, distinct geologic and climatic events that took place in or near the peninsula 78 have powered speciation, endemism, and separating distribution areas (Klimova et al., 2017). Three 79 of these major geological events are the formation of a midpeninsular seaway during middle 80 Pleistocene, a northward expansion of the Sea of Cortéz and the formation of a seaway across the 81 southern peninsular Isthmus of La Paz in the Pliocene (Riddle et al., 2000). Moreover, Baja 82 California has a complex tectonic history: the western portion is located on a transform fault, while the eastern part rests divergent boundary. Based on its geology and the biological communities 83 84 inhabiting it, eight biogeographic units were recently proposed within the peninsula, each one representing a unique biotic assemblage: Central Desert, Gulf Central Coast, Guadalupe Island, 85 86 Vizcaíno Desert, Sierra de la Giganta, Magdalena Plains, Cape and Sierra de la Laguna (Morrone, 87 2021).

88 Additionally, global events that took place during glaciations, such as changes in sea level, 89 global temperature, and precipitation (Hewitt, 2004) must be considered when inferring the 90 evolutionary history of the species. The Quaternary was a period characterized by multiple glacial 91 and interglacial cycles (Bennett, 1990; Mantooth et al., 2013). One of the most important events 92 occurred approximately 20,000 years ago during the Pleistocene: The Last Glacial Maximum (LGM), where glaciers around the world reached their maximum size. LGM the most impact near 93 94 the poles, where solar radiation is lowest, and regions experienced significant temperature decreases 95 that led to the expansion of glaciers. Temperate zones in the Northern hemisphere expanded 96 southwards towards the Equator, greatly shrinking tropical and template biomes (Williams et al., 2004). A particular case was that of the Western North American deserts. These ecosystems 97 98 experienced a reduction in extension during the Pleistocene glacial period, and it was not until after 99 the Wisconsin glaciation that the deserts reestablished completely, reaching its maximum range 100 during the Holocene Climatic Optimum (Mantooth et al., 2013; Riddle & Hafner, 2006).

101 In the same way as biomes, during the LGM many tropical and arid-related populations 102 went through demographic bottlenecks and fragmentations due to their habitats' reductions (Hewitt, 103 2000; Hewitt, 2004; Ony et al., 2021). In contrast, the interglacial period and the retraction of the 104 glaciers, those populations on the North hemisphere experienced a demographic growth and 105 geographic range expansion northwards (Nason et al., 2002). Species' genetic signatures of 106 divergence and subsequent diversification are concordant with the climatic and geographic 107 transformation of the region during the Quaternary in North America. Specifically, in species 108 associated with arid environments, there is evidence in their genomes of population differentiation 109 and isolation. This has been associated with populations distributed along Pleistocene refugia, 110 which later dispersed southwards after the LGM, expanding their distribution ranges, connectivity 111 and gene flow, as climatic conditions became more suitable and the vegetation recovered lost 112 ground (Graham et al., 2020; Sánchez-del Pino et al., 2020; Wilson & Pitts, 2012).

113 The combined use of genomic tools and species distribution models (SDM) allows us to 114 formally address biological questions from different perspectives, thus if both approaches agree, the 115 conclusions drawn gain more robustness (Aguirre-Liguori et al., 2021). Methods for estimating 116 effective population size from genomic data have had a big boom in diverse areas such as 117 evolutionary biology, ecology, and conservation genomics. Knowing genetic and demographic 118 patterns of populations in the past, might help predicting the population's behavior under future 119 scenarios like habitat loss or fragmentation, global warming, water scarcity, among others (Prates et 120 al., 2016). On the other hand, SDM enable to visualize changes in geographic space with respect to climate conditions and environmental resources necessary for the survival of species, associated 121 122 with their specific ecological niches (Miller, 2010). Thus, the joint study of distributional changes 123 and population demography can help understand more in-deep how global climatic changes have 124 shaped the populations we see today.

125	In this study we our aim was to model the effect of the climate changes that took place
126	during the LGM on the expansion of arid regions along the Baja California Peninsula, and to
127	determine if and how such habitat changes impacted the demography and genetic patterns of the
128	California gnatcatcher, Polioptila californica (Aves: Polioptilidae). P. californica is a small
129	perching bird endemic to the Baja California Peninsula, distributed in the arid and semi-arid regions
130	of southern California to the southernmost tip of Baja California Sur in Mexico. Previous studies
131	have shown a lack of genetic and geographic structure of this species along the peninsula as well as
132	first insights of a recent population expansion in concert with that matches the time following the
133	LGM (Vázquez-Miranda et al., 2022; Zink et al., 2013; Zink et al., 2016). These results suggest that
134	population maintained a continuous distribution or that the time elapsed since the genetic and
135	geographic isolation has not been enough for the subpopulations to diverge.
136	The infraspecific classification of the species is still in debate since multiple subspecies
137	hypothesis have been proposed (see Zink et al. 2000). Particularly, it has been suggested that the
138	populations inhabiting the coastal shrublands in south-eastern USA belong to a distinct taxonomic
139	entity, classified as the subspecies P. californica californica (McCormack & Maley, 2015). This
140	proposal is based on the amount of white color in the tail feathers, slightly sharper vocalizations and
141	habitat preferences (Atwood, 1988). This subspecies is listed in the Endangered Species Act (ESA;
142	16 USC 1531 et seq; U.S. Fish and Wildlife Services 1995) and is a flagship species for the
143	conservation of the coastal shrublands that inhabits. Those previous studies used a few
144	mitochondrial and nuclear genes. Therefore, assessing the evolutionary history of this species will
145	benefit from the analysis of multiple nuclear, unlinked loci. The lack of more comprehensive studies
146	makes the California gnatcatcher an ideal model to study the effect of climatic changes on the
147	genetic and geographic structure, demography, and distribution of peninsular species.
148	

150 **RESULTS**

151 Genetic diversity and population structure

Here we present the GBS-derived data for the estimation of genetic diversity and assessment 152 153 of population structure. From the VCF file we identified a total of 96,208 SNPs which was thinned 154 to keep only one SNP per locus resulting in 50,505 SNPs. After applying all the filters, the number 155 of SNPs was 9,233 for the whole date set, and 4,720 SNPs for the thinned one. Two individuals from the original sampling belonging to Cataviña population were removed from further analyses 156 157 since the preliminary analyses showed that the genetic differences between these two individuals and the remaining sampled individuals were greater than the differences to the sister species P. 158 159 melanura.

160 The estimated genetic diversity in *P. californica* was overall low. The mean observed 161 heterozygosity (H_o) among the five locations was 0.0661, while gene diversity was 0.0872. The 162 total inbreeding coefficient (F_{IS}) was 0.2411, although F_{IS} for any given location never exceeded 163 0.167. Similarly, the paired F_{ST} values were relatively low and never exceeded 0.072 indicating low 164 genetic differentiation between the sampled locations.

165 The low diversity and almost no differentiation was supported by the results obtained for genetic structure shown (Figure 1). The California gnatcatcher individuals clustered in a single 166 167 group. ADMIXTURE results found k=1 as the best model (for all 10 independent runs), based on 168 the smallest cross-validation error. Likewise, fastSTRUCTURE indicated that the most probable 169 number of populations ranged from one to two, which also suggests clustering of all the P. californica individuals as one genetic group. Similarly, DAPC results (Figure 1), showed 170 171 differences among groups are minor, with overlap between locations. Interestingly, only 172 gnatcatchers from Riverside and Santa Rosalillita have high assignment probabilities to their 173 nominal localities. The other samples have mixed probabilities that suggest lack of genetic 174 structure. The gene flow events also support the lack of structure (Figure 2), where the TreeMix

175 results showed that there was at least one migration event from El Cien in Baja California Sur to

176 Cataviña, Baja California. Even though the migration weight (i.e., the fraction of alleles in the target

177 population that came from the parental population) is low, it shows that gene flow did occur

178 between locations.



179

Fig. 1. Population genetic structure in the California gnatcatcher on GBS genomic data
inferred by different methods. A corresponds to the structure inferred with *ADMIXTURE* and B
with *fastSTRUCTURE*. The labels on top indicate the individuals' location. The color represents the
genetic cluster each individual was assigned to. C depicts the genetic clusters identified with the
first two linear discriminants (LD1 and LD2). The colors specify the location of each individual as
indicated in the bottom legend. RIV; Riverside, CAT; Cataviña, ROS; Santa Rosalillita, CIE; El
Cien, CAB; Cabo Pulmo, OUT; outgroup (*P. melanura*).

187

188 There is no clear clustering of the individuals based on the geographic location they were 189 collected from, except for the birds from Riverside and almost all the individuals from Santa 190 Rosalillita (Figure 3). The SVDquartets analysis revealed that birds from these two localities were 191 genetically closer to individuals from their same locality in comparison with the rest of the sampled





193

194 Fig. 2. Estimation of gene flow among *P. californica* populations with *TreeMix* with GBS data.

The yellow arrow represents an interchange of genetic alleles from the populations in El Cien to those in Cataviña. The x-axis indicates the amount of genetic drift experienced by each population, while the bar at the right illustrates the magnitude of the genetic flow. In this case only one event was inferred.

199



- 206 *melanura* individual belongs to the same genetic cluster as the focus species. The sister species was
- 207 placed at the center of all the *P. californica* individuals in the DAPC.



208

Fig. 3. Geographic structure of *P. californica* populations inferred with *SVDquartets* with GBS
data. In the right the code names for the sampled birds, whilst the colors denote the population each
individual belongs to. From top to bottom: black - OUT (outgroup), red - RIV (Riverside), purple ROS (Santa Rosalillita), green - CAB (Cabo Pulmo), yellow - CIE (El Cien), and blue - CAT
(Cataviña).

214

215 Historical demography

As runs reproduced the same demographic patterns independently of the number of free parameters (i.e., coalescent rates) indicated, we only describe the results from the runs with the lowest number of free parameters and, therefore, less associated error. The demographic shifts indicated by the red dark line in Figure 4 resume the MSMC2 results on RAD-Seq data using 18 free parameters (remaining runs on Supplementary Material). These revealed that *P. californica* had a high population size as far back as 100,000 years ago, with an estimated stable population size of up to a million effective birds. Around 18,000 years ago, *ca*. the end of the LGM (at the left of the dark blue block in Figure 4), the population size declined. The bottleneck was so strong that thepopulation effective size reached the tens of thousands.

The population decline continued reaching a thousand effective individuals, until about a hundred years ago that population size increased again reaching almost the same effective size as before the glacial period. These results suggest that the population has currently a stable size of hundreds of thousands. Although results were more consistent (bootstrap values) along the more recent times (Figure 4) they concurrently show that: (1) it was after the LGM peak that the population suffered the bottleneck, and (2) approximately 10,000 years ago the population expanded during the interglacial period.







autosomal RAD-Seq data. The dark red line depicts the results from the *MSCM2* run with the

original data, the translucent lines correspond to each one of the 20 runs made with bootstrap data.

236 The light blue rectangle indicates the time span of the Last Glacial Period, while the dark blue line

represents the time span for the LGM. Both axes are in log scale. LGM; Last Glacial Maximum,

238 Ne; population effective size.

Results for the sex Z chromosome using four parameters (Supplementary Material) showed that, even though the population size declined after the end of the glacial period, it was a gradual change (along about 10,000 years) in contrast with the autosomal analysis (Figure 5). Moreover, the main run and the replicates did not show the same patterns, although most of them exhibited a high population size. In addition, all bootstraps showed that the current population size is lower than that indicated by the main run. Nevertheless, all of them revealed markedly lower than the trend observed for autosomal SNPs, which may indicate that there is sex bias.



247

Fig. 5. Historical demographic changes in the sex Z chromosome of the California gnatcatcher estimated from RAD-Seq data. The dark red line depicts the results from the *MSCM2* run with the original data, the translucent lines correspond to each one of the 20 runs made with bootstrap data. The light blue rectangle indicates the time span of the Last Glacial Period, while the dark blue line represents the time span for the LGM. Both axis are in log scale. LGM; Last Glacial Maximum, Ne; population effective size.

254

255 We also evaluated the demographic history on the GBS data (Supplementary Material), 256 which showed almost the same demographic trends as the RAD-Seq data, with the difference that in 257 the former the population decline was not as fast. With these data population decline started about 258 10,000 years after the LGM ended, but less than five thousand years since the beginning of the 259 interglacial period. This trend continued until almost 100 years ago when the population increased. 260 Regarding sex chromosomes, results indicated a lack of agreement between the original run and the 261 bootstraps. Bootstraps indicated a high population size during the glacial period and a large part of the interglacial period, where the sex Z chromosome did not decrease in size until 1,000 years ago, 262 263 far from the LGM ending. This date is only a few thousand years after the date estimated with GBS 264 data.

265 Overall, both data show the same demographic patterns, differing only in the time where the 266 bottleneck began, with a signal of higher population size before the LGM, while after the glacial period ended populations went through a strong bottleneck. The estimated effective size for the sex 267 268 Z chromosome was lower compared with the autosomal chromosomes, with a longer time for the 269 sex chromosome to decline independently of the genomic data it was derived from.

270

Geographic distribution modelling

271 From a total of 1,065 occurrences (1,022 from GBIF, 38 from Waltari et al. 2007 and five 272 from our own records), cleaned occurrences resulted in only 269 samples. The collinearity test resulted in 10 non-correlated variables, including bio1 (annual mean temperature), bio2 (mean 273 274 diurnal range), bio4 (temperature seasonality), bio5 (max temperature of warmest month), bio8 275 (mean temperature of wettest quarter), bio9 (mean temperature of driest quarter), bio14 276 (precipitation of driest month), bio15 (precipitation seasonality), bio18 (precipitation of the warmest 277 quarter) and bio19 (precipitation of coldest quarter).

278 Results of the potential distribution model for the present, showed an AUC value of 0.965. 279 The predicted potential distribution for the California gnatcatcher included practically all the Baja 280 California Peninsula and the southern part of California (Figure 6). Despite the extent of the 281 distribution is high, the presence probability of the predicted area ranges from about 0.4 to 1, where 282 the coasts had higher probability. This can be associated with the fact that the Peninsula is 283 characterized by a high abundance of xeric scrubs and other arid ecosystems, which are the 284 preferred habitats of the species.



285

Fig. 6. Potential current distribution for *P. californica* inferred with *MaxEnt*. The color band at the right indicates the probability of occurrence of the species in the evaluated area. Warmer colors suggest higher probability. The size of each pixel is 2.5".

289

In contrast with the results for the present, both LGM models showed that the predicted extent of habitable area for *P. californica* was lower. However, the probability for the entire inferred areas was higher, ranging from 0.6 to 1. The latter suggest that in the LGM, there were more suitable places for the California gnatcatcher compared to those in the present (Figure 7). The LGM distribution showed that the species could not have lived in the northern region of the Peninsula neither in the south of California, where occurrence probability was almost zero. Thus, the habitable area in the LGM was restricted to mainly two regions: one to the southern Baja California Peninsula, again, particularly closer to the coast, and along the area joining the peninsula with the rest of the continent; a region that today is mostly below the sea.



Fig. 7. Potential distribution for *P. californica* for different historic times inferred with *MaxEnt*. A-B are the estimated potential distributions of the species during the LGM. A is based on
the MIROC general circulation model and B is based on the CCSM4 one. C is the potential
distribution during the LIG inferred with the only available global bioclimatic variables for this
period. The color band at the right indicates the probability of occurrence of the species in the
evaluated area. Warmer colors suggest higher probability. The size of each pixel is 2.5". LGM; Last
Glacial Maximum, LIG; Last Interglacial.

307

299

Regarding the potential distribution during the LIG (Last Interglacial), results showed that the predicted areas were located in the same region as in the LGM models (Figure 7). Nevertheless, the extent of both areas is smaller and the probability of occurrence is very low (ca. 40% presence). This indicates that the species distribution during the Interglacial was more restricted and climate conditions were not ideal.

313 **DISCUSSION**

314 Low genetic diversity is often seen in populations that have passed through a recent 315 bottleneck. In this scenario, the reduced population size would have increased the effect of genetic 316 drift leading to loss of random alleles, which in turn decreases genetic diversity. Additionally, 317 population size declining is linked with higher probabilities of inbreeding (Keller & Waller, 2002; 318 Palstra & Ruzzante, 2008; White & Searle, 2007). That seems to be the case for the California 319 gnatcatcher, which showed markedly low genetic variability. Zink et al. (2000) and Zink et al. 320 (2013) reported low genetic diversity for this species, particularly for populations distributed in the 321 north of the peninsula. We found lower values, which are the only ones based on multiple nuclear 322 loci. Furthermore, the overall inbreeding coefficients we observed are indicative of inbreeding 323 within the *P. californica* populations we studied.

324 Population bottlenecks and isolation have been associated with population structure, as shown by different species during the climatic oscillations of the Pleistocene (Burg et al., 2006; 325 McDonough et al., 2022; Menezes et al., 2021). However, our results consistently showed no 326 327 evidence of genetic population structure (F_{ST} and three population assignment methods). Only the 328 SVDquartets analysis suggested some potential grouping of individuals from Santa Rosalillita and Riverside. However, it is important to note that, because populations of this species from coastal 329 330 California are listed in the Endangered Species Act, we could not sample individuals and the three 331 ones used in this study came from a single nest with evidence of predation. Thus, it is not 332 unexpected that samples from Riverside are closely grouped.

Nonetheless, this was not the case for the Santa Rosalillita's samples, which given that it is located in the middle of the species range, it is unlikely this population diverged without the northern and southern populations diverging, particularly considering that there are no barriers isolating these individuals from the rest of the peninsula. The fact that we identified gene flow, suggesting that at least one event of genetic interchange between distant populations occurred,

makes it likely that gene flow happened between other populations. We acknowledge that we 338 339 needed a larger sample size to corroborate if this population is differentiated or if it was only an 340 artifact of our analysis. Nonetheless, our overall results support the hypothesis that *P. californica* is 341 not genetically structured, thus the proposed subspecies scheme is not adequate for this species. 342 Similar patterns have been observed for other bird species with comparable ecological 343 characteristics. Brown et al. (2013) found that the mallee emu-wren (Stipiturus mallee), an 344 Australian bird with low dispersal ability that inhabits semi-arid and fire-prone regions, has low 345 genetic diversity. This is due in part to the spatial and temporal changing patterns of its habitat, and because of the effects caused by fires that triggered local extinction and recolonization processes, 346 347 which in turn cause loss of genetic diversity and lack of genetic structure.

The historical demography and distribution patterns shown by the California gnatcatcher are 348 concordant with the above. The strong effect of the LGM climate changes is demonstrated by the 349 350 drastic population size decrement happening right after the end of the LGM. Moreover, the suggested potential distribution of *P. californica* during this period is restricted to the south. This 351 352 agrees with the south Pleistocene refugia hypothesis for species from the north hemisphere (Hewitt, 353 2001; Stewart et al., 2009). In fact, Waltari et al. (2007) found evidence indicating the existence of a P. californica refugium at the south of the peninsula. It has also been proposed as one of the main 354 355 phenomena that drove the distribution of other species, especially those with low dispersal 356 capability (Drovetski et al., 2018). Thus, the loss of suitable habitat in southern USA that took place during the LGM could have driven the decline of the *P. californica* populations. 357

When the glacial period ended, about 10,000 years ago, the range of the arid lands of North America expanded, enabling these ecosystems to reach distribution areas very similar to their present extent (Herzschuh, 2020; Moncrieff et al., 2015). Currently, the California gnatcatcher can be found as far as southern California and all along the Baja California Peninsula, supporting that the species' distribution could and did expand. However, although the availability of habitat

23

increased, it presumably took many generations for the species to recover. Two likely reasons for
the stable low effective sizes observed would be high genetic load and low genetic diversity. It is
not until the high inbreeding leads to the expression of deleterious mutations and natural selection
erases these mutations that populations can begin to grow and recover from the bottleneck (Jackson
et al., 2022).

368 Interestingly, our demographic and distribution models for the LIG (about 110,000 years 369 ago) are inconsistent. Namely, demographic results suggested there were millions of individuals, 370 whereas the SDM showed very low habitat suitability and availability. This conflicts with our 371 hypothesis, as we would expect low effective sizes associated with low habitat inferred area. 372 Nonetheless, it must be taken into account that the SDM models assume that ecological niches do not evolve, meaning that if the populations from which P. californica originated had a different 373 niche, it will not reflect the real biogeographic history of the species since the environmental 374 375 preferences considered were those of today's populations (Owens et al., 2013; Svenning et al., 376 2011). In addition, inference errors increase, and resolution power is lost as analysis pretend to 377 model distributions deeper into the past (Feng et al., 2019). Although the demographic method used 378 could also influence these results, we used MSMC2 that is a robust method when working with a higher number of haplotypes (Schiffels & Wang, 2020). 379

380 Despite the differences between the results obtained with GBS and with RAD-Seq data, the 381 overall historical demographic and genetic structure patterns are consistent. Both agree in terms of 382 the reduction of *P. californica* population size after the end of the LGM, only differing in the exact time to reach the bottleneck (by less than 10,000 years). They also concur in the lower effective size 383 384 estimated for sex chromosomes with respect of the autosomes, which might indicate there were 385 more females than males in the population, because in birds females are the heterogametic sex 386 (ZW) (Ellegren & Fridolfsson, 1997). This sex-ratio bias may influence genetic diversity and 387 population stability (Eberhart-Phillips et al., 2017). They also agree regarding the lack of genetic

structuring in *P. californica*. It should be mentioned that the grouping of *P. melanura* within a *P. californica* cluster is not due to a misidentification of the individual. This bird was collected from a
locality in Sonora, Mexico, which is not part of the California gnatcatcher natural distribution.
Thus, it could be the result of errors during the genomic library or the sequencing process.

We can conclude that our findings demonstrate that the geographic distribution and genetic patterns in the California gnatcatcher are tightly linked to the climatic and environmental changes during the Quaternary. Furthermore, the fact that *P. californica* responds in such tune with this habitat changes is of high concern under a scenario of global climatic change. It is still unknown if bird species with arid-like preferences will be favored by the contemporary desertification worldwide, but the rapid population's response shown in our study should be taken as a warning signal to target conservation efforts for this and other species with similar ecological niches.

399

400 MATERIALS AND METHODS

401 Sampling and genotyping

402 We used genomic data from Vazquez-Miranda et al. (2022; NCBI SRA BioProject 403 PRJNA719408). Briefly, those data consist of 23 individuals of P. californica obtained from five localities along the natural distribution of the species (Figure 1, Supplementary Material). DNA 404 405 extraction followed a phenol-chloroform-isoamyl protocol and two types of reduced-representation 406 sequencing (RRS) genomic libraries were created: one for single-end GBS (Genotyping By 407 Sequencing) and one for paired-end RAD-Seq (Restriction site Associated DNA Sequencing) data 408 following Gamble et al. (2015). For the GBS library, genomic DNA was digested with the PstI 409 enzyme, whereas for the RAD-Seq the *SbfI* enzyme library was used. Sequencing of both type of 410 libraries was carried out in independent lanes of the Illumina HiSeq 2000 platform at the University 411 of Minnesota. The individuals sequenced with RAD-Seq represent a subset of the individuals 412 sequenced with GBS. There were 23 California gnatcatcher samples for GBS data and 15 for RAD- 413 Seq. Additionally, in the sequencing one individual from the sister species, *P. melanura* (Zink &

Blackwell, 1998) was included, to enable structure comparisons and for rooting phylogeneticanalysis.

416 **Bioinformatics**

417 Raw reads were demultiplexed using Cutadapt (Martin, 2011) for the GBS data and PyRAD 418 (Eaton, 2014) for the RAD-Seq data. Then, adapters were removed with Cutadapt and reads were 419 trimmed with Trimommatic (Bolger et al., 2014). We evaluated the quality of the remaining reads 420 with FASTQC (Andrews, 2010) and created global reports with MultiQC (Ewels et al., 2016). 421 Additionally, we merged the R1 and R2 reads from RAD-Seq paired-end reads with PEAR (Zhang 422 et al., 2014). The cleaned reads were used as input for the BWA-MEM (Li, 2013) algorithm implemented in Galaxy (Afgan et al., 2018) to map against the nuclear genome of Taeniopygia 423 guttata (RefSeq GCF 003957565.2 bTaeGut1.4.pri). Next, we filtered the sequences that were not 424 425 a primary alignment or had a quality below 30 with BAMTools (Barnett et al., 2011). 426 We used the *ref map.pl* module from Stacks (Catchen et al., 2013) to identify the SNPs 427 based on the T. guttata nuclear genome. VCF files were generated using the populations module of 428 Stacks. For each genotyping method we had two VCF files: one with all the identified SNPs and

429 one with only one SNP per locus, the latter obtained by adding the argument *write-single-snp* from

430 *Stacks*. The final files were filtered with VCFTools (Danecek et al., 2011) so that we only recovered

431 biallelic loci, with a minimum allele count (MAC) of three or more, and up to 20% of missing data.

432 The software Plink (Purcell et al., 2007) was used to generate *bed*, *bim* and *treemix* files.

433 Genetic diversity and population structure

To estimate the genetic diversity, we used the VCF file that had all the SNPs, along with the R (R Core Team, 2022) *adegenet* (Jombart, 2008) and *hierfstat* (Goudet, 2005) packages. We estimated heterozygosity and gene diversity, as well as the paired fixation index (F_{ST}) following 438 evaluate genetic structure we applied three methods: ADMIXTURE (Alexander et al., 2009),

439 fastSTRUCTURE (Raj et al., 2014), and DAPC (Jombart et al., 2010), with which we evaluated 440 from k=2 (considering *P. californica* and *P. melanura* as two completely different genetic clusters) 441 to k=6 (deeming the individuals of each sampled locality as a distinct genetic group). Although the 442 three methods look for the number of clusters that best explain the genetic variation in the data, 443 each one follows a slightly different approach. ADMIXTURE is based on maximum likelihood, 444 fastSTRUCTURE works with Bayesian inference, and DAPC is a multivariate approach which does 445 not consider the nominal populations nor the identity of the individuals (Stift et al., 2019). We ran 10 repetitions on each k on each approach. 446

447 To determine the geographic structure of the California Gnatcatcher, we inferred an individual-level phylogeny. We achieved this with SVDquartets (Chifman & Kubatko, 2014) 448 implemented in PAUP* (Swofford, 2003). Since this program requires a NEXUS file, we first used 449 the Python script vcf2fasta.pv (https://github.com/santiagosnchez/vcf2fasta) to obtain a fasta file 450 from the VCF file that accounts for linkage disequilibrium. Then, we transformed the fasta file into 451 452 NEXUS using ALTER (Glez-Peña et al., 2010). Finally, we ran SVDquartets and calculated the 453 bootstrap value for each branch. These results were plotted on a densitree using ape (Paradis & Schliep, 2019) and phangorn (Schliep, 2011). 454

Also, we estimated gene flow with TreeMix (Pickrell & Pritchard, 2012), which works with a maximum likelihood approximation. For this we tried up to eight migration events (*M*) and ran each *M* 10 independent times. To determine the number of migratory events that maximizes the likelihood of the model we used OptM (Fitak, 2021) and *popcorn*

459 (https://github.com/andrewparkermorgan/popcorn) to plot the resulting graph.

460 Historical demography

461 To evaluate changes in effective population size we first used clean and trimmed reads from 462 one individual from each sampled location to reduce the computational time. From these reads we 463 removed every read that mapped against the mitochondrial genome of *T. guttata* (RefSeq

464 NC_007897.1), and then mapped to the nuclear genome of the Zebra Finch. These reads were then
465 passed through *MarkDuplicates* in Picard (http://broadinstitute.github.io/picard/) implemented in
466 GATK (van der Auwera & O'Connor, 2020) to remove sequence duplicates.

467 After this, we followed the instructions indicated by the authors of MSMC (Schiffels & 468 Durbin, 2014) and called the SNPs in a chromosome-based fashion with the tool mpileup from 469 SAMTools. Then, by combining the data from all the individuals, we generated the *multihetsep* files 470 with the 'generate multihetsep.py' tool (https://github.com/stschiff/msmc-tools). These files have 471 four columns: one with the number of the chromosome, one with the particular position on that 472 chromosome, one with the total number of homozygous sites since the last segregation site, and the last one with the called alleles fort the heterozygous sites, in order of appearance, of the evaluated 473 haplotypes. We did not phase the SNPs nor did we used a mask file, since we used MSMC2, rather 474 475 than MSMC, as we only wanted to estimate the effective size changes and not the divergence time 476 (Schiffels & Wang, 2020). Two analyses with different sets of *multihetsep* files were ran: one with 477 all the autosomal chromosomes and one for the Z chromosome only.

478 With this *multihetsep* files we ran the program *msmc2* (https://github.com/stschiff/msmctools) and evaluated up to three different segmentation patterns using 18, 20, and 28 free parameters 479 480 for the autosomal chromosomes, and four, four, six, and seven parameters for the sexual 481 chromosome Z. Additionally, the program was forced to run only on specified pairs. This assumes 482 that each indicated pair of haplotypes belongs to a different population. Lastly, we used the 'multihetsep bootstrap.py' to create 20 sets of multihetsep files with random regions of missing 483 484 data to evaluate the robustness of the analysis. In each of these sets we re-ran msmc2 under the 485 same parameters. The bootstrap was only estimated for the 18- and the four-parameters runs.

486 To scale the data to individuals per year, we used the mutational rate per year in

487 Passeriformes (Smeds et al., 2016), which is 2.3 x 10⁻⁹, as well as the generation time which in *P*.
488 *caerulea* is estimated to be of one year (Smith et al., 2018).

489 Geographic distribution modelling

490 First, we downloaded the occurrence data of *P. californica* from GBIF, using *rgbif* 491 (Chamberlain et al., 2017). To reduce potential errors, we only used those records based on 492 preserved specimens that have associated coordinates. The occurrences were filtered with 493 CoordinateCleaner (Zizka et al., 2019) and spThin (Aiello-Lammens et al., 2015) to remove duplicated data and to thin the registers using a distance buffer of 2 km, respectively. Also, we 494 495 added five occurrence records that correspond to the five locations we sampled to obtain the genomic data and the 39 occurrences used by Waltari et al. (2007) to identify Pleistocene refugia for 496 the California gnatcatcher. 497

To select the environmental data for our model, we used the 19 bioclimatic variables of WordlClim2 (Fick & Hijmans, 2017) from 1970-2000 at a resolution of 2.5 minutes. These files were cropped in QGIS (QGIS.org, 2022) using a squared *shape* file so that the entire actual extent of the Baja California Peninsula was included. Additionally, we downloaded and cropped the 19 bioclimatic variables for the LGM of two different General Circulation Models (CGM): CCSM4 y MIROC-ESM (Fick & Hijmans, 2017), and the ones for the Last Interglacial. All at the same resolution of 2.5 minutes.

To create the model, we tested for collinearity among the current variables with the function *vifcor* from the R package *usdm* (Naimi et al., 2014) with a threshold value of 0.8. The variables with the highest VIF were excluded. Using the remaining variables and the final occurrence database, we created a model of the potential geographic distribution using the MaxEnt algorithm (Phillips et al., 2006) implemented in *dismo* (Hijmans et al., 2015). To build the potential distribution model for the present, we randomly kept 20% of the total occurrences to evaluate the

511	model. We used the linear, quadratic and product functions in MaxEnt considering that the shape of
512	an ecological niche is ellipsoidal and other functions add unnecessary complexity and reduce the
513	performance of the algorithm (Warren & Seifert, 2011). We then projected this model to the three
514	past scenarios: the LGM model CCSM4, the LGM MIROC-ESM one, and the LIG (Last
515	Interglacial). Finally, to evaluate the differences among the created models, we used ENMTools
516	(Warren et al., 2021) to calculate the area of overlap between the contemporary and the past models.
517	
518	DATA ACCESSIBILITY
519	The datasets supporting this article, as well as the scripts used for the analyses, were
520	deposited in https://figshare.com/s/d8842f8a34e7d486bcdc.
521	
522	FUNDING
523	CM-A was funded by a Consejo Nacional de Ciencia y Tecnología (CONACyT) masters
524	fellowship (CVU 1101167). HV-M was supported by two Universidad Nacional Autónoma de
525	México (UNAM) DGAPA-PAPIIT (IA204220 & IA205422) grants.
526	
527	ACKNOWLEDGEMENTS
528	This paper serves as a fulfillment of CM-A for obtaining a M.Sc. degree in the Posgrado en
529	Ciencias Biológicas, UNAM. We thank the Instituto de Biología and Posgrado en Ciencias
530	Biológicas at UNAM for providing the support and infrastructure that made possible this project, as
531	well as CONACyT for the fellowship given to CM-A.
532	
533	REFERENCES
534	Afgan E et al. 2018. The Galaxy platform for accessible, reproducible and collaborative biomedical
535	analyses: 2018 update. Nucleic Acids Res. 46(W1):W537-W544.

- Aguirre-Liguori JA, Ramírez-Barahona S, Gaut BS. 2021. The evolutionary genomics of species'
 responses to climate change. Nat. Ecol. Evol. 5(10):1350–1360.
- 538 Aiello-Lammens ME, Boria RA, Radosavljevic A, Vilela B, Anderson RP. 2015. spThin: an R
- 539 package for spatial thinning of species occurrence records for use in ecological niche models.
- 540 Ecography 38(5):541–545.
- Alexander DH, Novembre J, Lange K. 2009. Fast model-based estimation of ancestry in unrelated
 individuals. Genome Res. 19(9):1655–1664.
- Álvarez-Castañeda ST, Murphy RW. 2014. The endemic insular and peninsular species *Chaetodipus spinatus* (mammalia, heteromyidae) breaks patterns for Baja California. PLoS ONE 9(12):1–
- 545 26.
- 546 Andrews S. 2010. FastQC A Quality Control tool for High Throughput Sequence Data.
- 547 https://www.bioinformatics.babraham.ac.uk/projects/fastqc/ (February 23th, 2023).
- 548 Atwood JL. 1988. Speciation and Geographic Variation in Black-Tailed Gnatcatchers. Ornithol.
 549 Monogr. 42:iii–74.
- Barnett DW, Garrison EK, Quinlan AR, St\u00efmberg MP, Marth GT. 2011. BamTools: a C++ API and
 toolkit for analyzing and managing BAM files. Bioinformatics 27(12):1691–1692.
- Bennett KD. 1990. Milankovitch Cycles and Their Effects on Species in Ecological and
 Evolutionary Time. Paleobiology 16(1):11–21.
- Bolger AM, Lohse M, Usadel B. 2014. Trimmomatic: A flexible trimmer for Illumina sequence
 data. Bioinformatics 30(15):2114–2120.
- Brown JW. 1987. The Peninsular Effect in Baja California: An Entomological Assessment. J.
 Biogeogr. 14(4):359.
- 558 Brown SM, Harrisson KA, Clarke RH, Bennett AF, Sunnucks P. 2013. Limited Population
- 559 Structure, Genetic Drift and Bottlenecks Characterise an Endangered Bird Species in a
- 560 Dynamic, Fire-Prone Ecosystem. PLoS ONE 8(4):e59732.

- Burg TM, Gaston AJ, Winker K, Friesen VL. 2006. Effects of Pleistocene glaciations on population
 structure of North American chestnut-backed chickadees. Mol. Ecol. 15(9):2409-2419.
- 563 Catchen JM, Hohenlohe PA, Bassham S, Amores A, Cresko WA. 2013. Stacks: an analysis tool set
 564 for population genomics. Mol. Ecol. 22(11):3124–3140.
- 565 Chamberlain S, Oldoni D, Waller J. 2017. rgbif: interface to the global biodiversity information
- 566 facility API. https://cran.r-project.org/web/packages/rgbif/index.html (February 23th, 2023).
- 567 Chifman J, Kubatko L. 2014. Phylogenetics Quartet Inference from SNP Data Under the Coalescent
 568 Model. Bioinformatics 30(23):3317–3324.
- 569 Danecek P et al. 2011. The variant call format and VCFtools. Bioinformatics 27(15):2156–2158.
- 570 Drovetski Sv et al. 2018. A test of the European Pleistocene refugial paradigm, using a Western
- 571 Palaearctic endemic bird species. Proc. R. Soc. B: Biol. Sci. 285(1889):20181606.
- Eaton DAR. 2014. PyRAD: Assembly of de novo RADseq loci for phylogenetic analyses.
 Bioinformatics 30(13):1844–1849.
- Eberhart-Phillips LJ et al. 2017. Sex-specific early survival drives adult sex ratio bias in snowy
 plovers and impacts mating system and population growth. *Proc. Natl. Acad. Sci. U.S.A.*114(27):E5474–E5481.
- 577 Ellegren H, Fridolfsson A-K. 1997. Male–driven evolution of DNA sequences in birds. Nat. Genet.
 578 17(2):182–184.
- Ewels P, Magnusson M, Lundin S, Käller M. 2016. MultiQC: summarize analysis results for
 multiple tools and samples in a single report. Bioinformatics 32(19):3047–3048.
- Feng X, Park DS, Liang Y, Pandey R, Papeş M. 2019. Collinearity in ecological niche modeling:
 Confusions and challenges. Ecol. Evol. 9(18):10365–10376.
- Fick SE, Hijmans RJ. 2017. WorldClim 2: new 1-km spatial resolution climate surfaces for global
 land areas. Int. J. Climatol. 37(12):4302–4315.

- Fitak RR. 2021. OptM : estimating the optimal number of migration edges on population trees using
 Treemix. Biol. Methods Protoc. 6(1):1–6.
- Gamble T, Coryell J, Lynch J, Scantlebury PD, Zarkower D. 2015. Restriction site- associated DNA
 sequencing (RAD-seq) reveals an extraordinary number of transitions among gecko sexdetermining systems. Mol. Biol. Evol. 32:1296–1309.
- 590 Glez-Peña D, Gómez-Blanco D, Reboiro-Jato M, Fdez-Riverola F, Posada D. 2010. ALTER:
- 591 program-oriented conversion of DNA and protein alignments. Nucleic Acids Res.
- 592 38(suppl_2):W14–W18.
- Goudet J. 2005. HIERFSTAT, a package for R to compute and test hierarchical F-statistics. Mol.
 Ecol. Notes 5(1):184–186.
- 595 Graham MR, Santibáñez-López CE, Derkarabetian S, Hendrixson BE. 2020. Pleistocene
- persistence and expansion in tarantulas on the Colorado Plateau and the effects of missing data
 on phylogeographical inferences from RADseq. Mol. Ecol. 29(19):3684–3701.
- Herzschuh U. 2020. Legacy of the Last Glacial on the present-day distribution of deciduous versus
 evergreen boreal forests. Glob. Ecol. Biogeogr. 29(2):198–206.
- Hewitt GM. 2000. The genetic legacy of the Quaternary ice ages. Nature 405(6789):907-913.
- Hewitt GM. 2001. Speciation, hybrid zones and phylogeography or seeing genes in space and
 time. Mol. Ecol. 10(3):537–549.
- Hewitt GM. 2004. Genetic consequences of climatic oscillations in the Quaternary. *Philos. Trans. R. Soc.* B: Biol. Sci. 359(1442):183–195.
- 605 Hijmans RJ, Phillips S, Leathwick J, Elith J. 2015. Dismo: Species Distribution Modeling.
- 606 https://cran.r-project.org/web/packages/dismo/index.html (February 23th, 2023).
- Hoskin CJ, Higgie M, McDonald KR, Moritz C. 2005. Reinforcement drives rapid allopatric
 speciation. Nature 437(7063):1353–1356.

- Jackson HA et al. 2022. Genomic erosion in a demographically recovered bird species during
 conservation rescue. Conserv. Biol. 36(4):e13918.
- Jombart T. 2008. Genetics and population analysis adegenet: a R package for the multivariate
 analysis of genetic markers. Bioinformatics 24(11):1403–1405.
- Jombart T, Devillard S, Balloux F. 2010. Discriminant analysis of principal components: A new
 method for the analysis of genetically structured populations. BMC Genet. 11(94).
- Keller LF, Waller DM. 2002. Inbreeding effects in wild populations. Trends Ecol. Evol. 17(5):230–
 241.
- 617 Klimova A, Hoffman JI, Gutierrez-Rivera JN, Leon de la Luz J, Ortega-Rubio A. 2017. Molecular
- 618 genetic analysis of two native desert palm genera, *Washingtonia* and *Brahea*, from the Baja
- 619 California Peninsula and Guadalupe Island. Ecol. Evol. 7(13):4919–4935.
- Klinga P et al. 2019. Considering landscape connectivity and gene flow in the Anthropocene using
 complementary landscape genetics and habitat modelling approaches. Landsc. Ecol.
- 622 34(3):521–536.
- Leaché AD, Crews SC, Hickerson MJ. 2007. Two waves of diversification in mammals and reptiles
 of Baja California revealed by hierarchical Bayesian analysis. Biol. Let. 3(6):646–650.
- 625 Li H. 2013. Aligning sequence reads, clone sequences and assembly contigs with BWA-MEM.
- 626 BioRxiv arXiv:1303.3997v2 [q-bio.GN]
- Mantooth SJ, Hafner DJ, Bryson RW, Riddle BR. 2013. Phylogeographic diversification of antelope
 squirrels (*Ammospermophilus*) across North American deserts. Biol. J. Linn. Soc. 109(4):949–
 967.
- 630 Martin M. 2011. Cutadapt removes adapter sequences from high-throughput sequencing reads.
- 631 EMBnet j. 17(1):10.

- McCormack JE, Maley JM. 2015. Interpreting negative results with taxonomic and conservation
 implications: Another look at the distinctness of coastal California Gnatcatchers. Auk *132*(2):380–388.
- McDonough MM, Ferguson AW, Dowler RC, Gompper ME, Maldonado JE. 2022. Phylogenomic
 systematics of the spotted skunks (Carnivora, Mephitidae, *Spilogale*): additional species
- 637 diversity and Pleistocene climate change as a major driver of diversification. Mol. Phylogenet.
 638 Evol. 167:107266.
- 639 Menezes L, Batalha-Filho H, Garda AA, Felgueiras Napoli M. 2021. Tiny treefrogs in the
- 640 Pleistocene: Phylogeography of *Dendropsophus oliveirai* in the Atlantic Forest and associated
 641 enclaves in northeastern Brazil. J. Zoolog. Syst. Evol. Res. 59(1):179-194.
- 642 Miller J. 2010. Species distribution modeling. Geogr. Compass 4(6):490–509.
- Moncrieff GR, Hickler T, Higgins SI. 2015. Intercontinental divergence in the climate envelope of
 major plant biomes. Glob. Ecol. Biogeogr. 24(3):324–334.
- Morrone JJ. 2021. Biogeographic regionalisation of the Baja California biogeographic province,
 Mexico: A review. J. Nat. Hist. 55(5-6):365–379.
- Naimi B, Hamm NAS, Groen TA, Skidmore AK, Toxopeus AG. 2014. Where is positional
 uncertainty a problem for species distribution modelling? Ecography 37(2):191–203.
- 649 Nason JD, Hamrick JL, Fleming TH. 2002. Historical vicariance and postglacial colonization
- 650 effects on the evolution of genetic structure in Lophocereus, a Sonoran desert columnar cactus.
- 651 Evolution 56(11):2214–2226.
- National Research Council. 1995. *Science and the Endangered Species Act*. Washington, DC: The
 National Academies Press.
- Ony M et al. 2021. Genetic diversity in North American Cercis canadensis reveals an ancient
 population bottleneck that originated after the last glacial maximum. Sci. Rep. 11:21803.

- 656 Owens HL et al. 2013. Constraints on interpretation of ecological niche models by limited
- 657 environmental ranges on calibration areas. Ecol. Modell. 263:10–18.
- 658 Palstra FP, Ruzzante DE. 2008. Genetic estimates of contemporary effective population size: what
- 659 can they tell us about the importance of genetic stochasticity for wild population persistence?
- 660 Mol. Ecol. 17(15):3428–3447.
- Paradis E, Schliep K. 2019. ape 5.0: an environment for modern phylogenetics and evolutionary
 analyses in R. Bioinformatics 35(3):526–528.
- Phillips SJ, Anderson RP, Schapire RE. 2006. Maximum entropy modeling of species geographic
 distributions. Ecol. Modell. 190(3–4):231–259.
- 665 Pickrell JK, Pritchard JK. 2012. Inference of Population Splits and Mixtures from Genome-Wide
- Allele Frequency Data. PLoS Genetics 8(11):e1002967.
- Prates I et al. 2016. Inferring responses to climate dynamics from historical demography in
 neotropical forest lizards. *Proc. Natl. Acad. Sci. U.S.A.* 113(29):7978–7985.
- Purcell S et al. 2007. PLINK: A tool set for whole-genome association and population-based
 linkage analyses. *Am. J. Hum. Genet.* 81(3):559–575.
- 671 Pyron RA, Burbrink FT. 2010. Hard and soft allopatry: Physically and ecologically mediated modes
 672 of geographic speciation. J. Biogeogr. 37(10):2005–2015.
- 673 QGIS.org. 2022. QGIS Geographic Information System. QGIS Association. http://www.qgis.org
 674 (February 23th, 2023).
- R Core Team. 2022. The R Project for Statistical Computing. https://www.r-project.org/ (February
 23th, 2023).
- 677 Raj A, Stephens M, Pritchard JK. 2014. fastSTRUCTURE: Variational Inference of Population
- 678 Structure in Large SNP Data Sets. Genetics 197(2):573–589.

- 679 Riddle BR, Hafner DJ, Alexander LF, Jaeger JR. 2000. Cryptic vicariance in the historical assembly
- of a Baja California Peninsular Desert biota. Proc. Natl. Acad. Sci. U.S.A. 97(26):14438–
 14443.
- Riddle BR, Hafner DJ. 2006. Biogeografía histórica de los desiertos cálidos de norteamérica. In:
- 683 Vázquez-Domínguez E, Hafner DJ, editors. *Genética y mamíferos mexicanos: presente y*
- *futuro: Bulletin 32*, 32. New Mexico: New Mexico Museum of Natural History and Science
 Bulletin. pp. 57–65.
- Sánchez-del Pino, I. et al. (2020). High phylogeographic and genetic diversity of *Tidestromia lanuginosa* supports full-glacial refugia for arid-adapted plants in southern and central
 Coahuila, Mexico. *Am. J. Bot.* 107(9):1296–1308.
- Schiffels, S., & Durbin, R. (2014). Inferring human population size and separation history from
 multiple genome sequences. Nat. Genet. 46(8):919–925.
- 691 Schiffels S, Wang K. 2020. MSMC and MSMC2: The Multiple Sequentially Markovian Coalescent.
- 692 In: Dutheil JY, editor. *Statistical Population Genomics. Methods in Molecular Biology*, 2090.
- 693 New York: Humana. pp. 147–166.
- 694 Schliep KP. 2011. phangorn: phylogenetic analysis in R. Bioinformatics 27(4):592–593.
- 695 Sedlock RL. 2003. Geology and tectonics of the Baja California Peninsula and adjacent areas. In:
- 696Johnson SE et al, editors. Special Paper 374: Tectonic evolution of northwestern Mexico and
- 697 *the Southwestern USA*, 374. Colorado: Geological Society of America. pp. 1–42.
- 698 Smeds L, Qvarnström A, Ellegren H. 2016. Direct estimate of the rate of germline mutation in a
- 699 bird. Genome Res. 26(9):1211–1218.
- Smith BT et al. 2018. Species delimitation and biogeography of the gnatcatchers and gnatwrens
 (Aves: Polioptilidae). Mol. Phylogenet. Evol. 126(January):45–57.
- 702 Stewart JR, Lister AM, Barnes I, Dalen L. 2009. Refugia revisited: individualistic responses of
- 703 species in space and time. Proc. R. Soc. B: Biol. Sci. 277(1682):661–671.

- Stift M, Kolář F, Meirmans PG. 2019. Structure is more robust than other clustering methods in
 simulated mixed-ploidy populations. Heredity (Edinb) 123(4):429–441.
- Svenning J-C, Fløjgaard C, Marske KA, Nógues-Bravo D, Normand S. 2011. Applications of
 species distribution modeling to paleobiology. *Quat. Sci. Rev.* 30(21–22):2930–2947.
- 708 Swofford DL. 2003. *PAUP**. *Phylogenetic analysis using parsimony (* and other methods)*.
- 709 Massachusetts: Sinauer Associates.
- van der Auwera GA, O'Connor BD. 2020. *Genomics in the Cloud: Using Docker, GATK, and WDL in Terra*. California: O'Reilly Media.
- 712 Vázquez-Miranda H, Zink RM, Pinto BJ. 2022. Comparative phylogenomic patterns in the Baja
- 713 California avifauna, their conservation implications, and the stages in lineage divergence. Mol.
- 714 Phylogenet. Evol. 171:107466.
- Waltari E, Hijmans RJ, Peterson AT, Nyári AS, Perkins SL. 2007. Locating Pleistocene Refugia:
 Comparing Phylogeographic and Ecological Niche Model Predictions. PLoS ONE 2(7):563.
- 717 Warren DL, Seifert SN. 2011. Ecological niche modeling in Maxent: The importance of model
- complexity and the performance of model selection criteria. Ecol. Appl. 21(2):335–342.
- Warren DL et al. 2021. ENMTools 1.0: an R package for comparative ecological biogeography.
 Ecography 44(4):504–511.
- Weir BS, Cockerham CC. 1984. Estimating F-Statistics for the Analysis of Population Structure.
 Evolution 38(6):1358–1370.
- White TA, Searle JB. 2007. Genetic diversity and population size: Island populations of the
 common shrew, *Sorex araneus*. Mol. Ecol. 16(10):2005–2016.
- Williams JW, Shuman BN, Webb T, Bartlein PJ, Leduc PL. 2004. Late-Quaternary vegetation
 dynamics in North America: Scaling from taxa to biomes. Ecol. Monogr. 74(2):309–334.
- 727 Wilson JS, Pitts JP. 2012. Identifying Pleistocene refugia in North American cold deserts using
- phylogeographic analyses and ecological niche modelling. Diver. Distrib. 18(11):1139–1152.

- Zhang J, Kobert K, Flouri T, Stamatakis A. 2014. PEAR: A fast and accurate Illumina Paired-End
 reAd mergeR. Bioinformatics 30(5):614–620.
- 731 Zink RM, Blackwell RC. 1998. Molecular Systematics and Biogeography of Aridland Gnatcatchers
- (Genus *Polioptila*) and Evidence Supporting Species Status of the California Gnatcatcher
 (*Polioptila californica*). *Mol. Phylogenet. Evol.* 9(1):26–32.
- 734 Zink RM, Barrowclough GF, Atwood JL, Blackwell-Rago RC. 2000. Genetics, taxonomy, and
- conservation of the threatened California Gnatcatcher. Conserv. Biol. 14(5):1394–1405.
- 736 Zink RM, Groth JG, Vazquez-Miranda H, Barrowclough GF. 2013. Phylogeography of the
- 737 California Gnatcatcher (Polioptila californica) using multilocus DNA sequences and
- ecological niche modeling: Implications for conservation. Auk 130(3):449–458.
- 739 Zink RM, Groth JG, Vázquez-Miranda H, Barrowclough GF. 2016. Geographic variation, null
- hypotheses, and subspecies limits in the California Gnatcatcher: A response to McCormack
 and Maley. Auk 133(1):59–68.
- 742 Zizka A et al. 2019. CoordinateCleaner: Standardized cleaning of occurrence records from
- 543 biological collection databases. Methods Ecol. Evol. 10(5):744–751.

Discusión y conclusiones

En poblaciones que recientemente han experimentado cuellos de botella se suele observar una disminución del tamaño poblacional, así como una baja diversidad genética (Forsdick et al., 2017; Hu et al., 2020). La reducción del tamaño poblacional resultado del cuello de botella incrementa el efecto de la deriva génica, con consecuente pérdida aleatoria de alelos y disminución de la diversidad genética. Además, se ha demostrado que la reducción poblacional se relaciona con una mayor probabilidad de apareamientos no aleatorios, aumentando la consanguinidad (Keller y Waller, 2002; Palstra y Ruzzante, 2008; White y Searle, 2007). Este parece ser el caso de la perlita de California, sobre la cual Zink et al. (2000) y Zink et al. (2013) reportaron baja diversidad genética, particularmente para las poblaciones del norte de la península. Aquí, usando múltiples loci nucleares, encontramos valores de diversidad genética más bajos a los reportados en estudios previos. Asimismo, los valores de endogamia que observamos indican consanguinidad moderada en la población estudiada.

Los cuellos de botella y el aislamiento poblacional se han relacionado con la estructura poblacional, como se ha visto previamente con otras especies durante las oscilaciones climáticas del Pleistoceno (Burg et al., 2006; McDonough et al., 2022; Menezes et al., 2021). En el caso de la perlita de California, los resultados mostraron de manera consistente que no existe evidencia de estructura genética (F_{ST} y tres métodos de asignación poblacional). Solamente el análisis con SVDquartets sugirió un potencial agrupamiento de los individuos de Santa Rosalillita y los de Riverside; sin embargo, es necesario mencionar que, ya que las poblaciones de esta ave que habitan en la costa de California forman parte del Acta de Especies en Peligro de Extinción de EE.UU., no fue posible muestrear individuos en California. De hecho, las tres muestras de Riverside que empleamos en este estudio provienen de un único nido encontrado en campo, mismo que tenía indicios de depredación. Por ello, no resulta extraño que los individuos de dicha localidad formen un clúster genético.

Este no fue el caso de las muestras de Santa Rosalillita donde, debido a que la localidad se encuentra en la parte media del área de distribución de la especie, es poco probable que estos individuos hubieran divergido sin que, de igual forma, lo hicieran las poblaciones del norte y el sur; especialmente considerando que no existen barreras que aíslen a esta localidad del resto de la península. El hecho de que los resultados indicaran la presencia de flujo génico, donde encontramos soporte para al menos un evento de intercambio genético entre localidades distantes, sugiere que puede existir flujo génico entre otras localidades. Estudios futuros podrían ampliar el muestreo para genéticamente del resto o es un artefacto del diseño de nuestro estudio. No obstante, en general, los resultados aquí obtenidos apoyan la hipótesis de que *P. californica* no presenta estructura genética, por lo que cualquier esquema de clasificación infraespecífico resulta inadecuado para la historia evolutiva de la especie.

De manera importante, se han reportado resultados similares en otra especie con características ecológicas similares. Brown et al. (2013) encontraron que el maluro de mallee (*Stipiturus mallee*), un ave australiana con baja capacidad de dispersión que habita en ambientes semiáridos propensos a incendios presenta baja diversidad genética. Esto es en parte debido a los patrones espaciotemporales cambiantes de su hábitat, pero también debido a procesos de extinción y recolonización locales dados por los incendios. Mismos que, a su vez, provocan la pérdida de diversidad y estructura genética.

Estos patrones genéticos concuerdan con los resultados de demografía histórica y distribución geográfica. En primer lugar, que la especie experimente un declive demográfico hacia finales del Último Glacial Máximo (UMG) muestra algunos de los potenciales efectos de los cambios ambientales sobre la especie. Además, la distribución potencial inferida con los SDM para la perlita de California durante este periodo indica que la especie se encontraba restringida al sur de la península, región que ya había sido propuesta como un refugio pleistocénico para otras especies del Hemisferio Norte (Hewitt, 2001; Stewart et al., 2009). De hecho, Waltari et al. (2007) encontraron evidencias de poblaciones aisladas de *P. californica* en la porción sur de la península. Asimismo, se ha propuesto que la recolonización desde refugios pleistocénicos es uno de los principales fenómenos determinantes asociados con la distribución actual de muchas especies, especialmente para aquellas con baja capacidad de dispersión (Drovetski et al., 2018). Así, la reducción en el tamaño efectivo de las poblaciones de *P. californica* se relaciona directamente con la pérdida de hábitat adecuado al sur de EE.UU. que ocurrió durante el UMG.

Al término del periodo glacial, hace aproximadamente 10,000 años, la temperatura global ascendió y las tierras áridas de Norteamérica pudieron extenderse hacia el norte, permitiendo que los ecosistemas alcanzaran una distribución muy similar a la actual (Herzschuh, 2020; Moncrieff et al., 2015). Actualmente, la perlita de California se distribuye en toda la península de Baja California, desde el sur de California en EE.UU. hasta la punta sur de Baja California Sur en México. Dada la diferencia entre la distribución durante el UMG y la distribución actual, inferimos que la distribución de la especie efectivamente se incrementó. Sin embargo, a pesar de una mayor disponibilidad de hábitat, tuvieron que transcurrir múltiples generaciones para que la especie pudiera recuperarse en términos demográficos. Dos posibles explicaciones para la estabilidad del reducido tamaño efectivo de la especie son la alta carga genética y la baja diversidad genética, donde es necesario que existan

altos niveles de consanguinidad en una especie para inducir la expresión de mutaciones deletéreas. Es entonces cuando la selección natural pudiera actuar eliminando dichas mutaciones, permitiendo que la población crezca y salga del cuello de botella (Jackson et al., 2022).

De la misma forma, cerca del Último Interglacial (UIG), hace aproximadamente 110,000 años, existe una contradicción entre los resultados de los análisis demográficos y los geográficos. Por una parte, la demografía indica que en el periodo interglacial la población alcanzaba un tamaño efectivo que rondaba los millones de individuos, mientras que los análisis geográficos demuestran que la idoneidad y disponibilidad del ambiente eran bajas. Esto último está en contraposición con nuestra hipótesis, ya que esperaríamos que la inferencia de áreas habitables reducidas se asocie con tamaños efectivos reducidos. Sin embargo, es importante tomar en cuenta que los modelos SDM asumen que el nicho ecológico de las especies no cambia, es decir, que el nicho no evoluciona. Lo que implica que si la población que dio origen a P. californica tenía preferencias ambientales diferentes a las de la población actual entonces el modelo no reflejaría de manera certera la distribución de la población durante el UIG (Owens et al., 2013; Svenning et al., 2011). Debemos considerar que este tipo de modelos pierden resolución conforme nos alejamos demasiado en el tiempo, ya que el error de inferencia incrementa en tiempos más antiguos (Feng et al., 2019). Esta falta de concordancia también puede deberse al modelo demográfico empleado, sin embargo, en este estudio usamos el programa MSMC2, el cual es más robusto a tiempos antiguos incluso cuando se emplea con un número de haplotipos relativamente alto (Schiffels y Wang, 2020).

A pesar de las diferencias observadas entre los resultados obtenidos con GBS (Genotipado por secuenciación) y con RAD-Seq (Secuenciación de DNA asociada a sitios de restricción), los patrones generales de demografía histórica son consistentes. Ambos concuerdan en la reducción del tamaño efectivo de *P. californica* al término del UMG, aunque difieren (por menos de 10,000 años) en el tiempo exacto en el cual la especie entró en el cuello de botella. También concuerdan en el bajo tamaño efectivo estimado para los cromosomas sexuales con respecto al estimado para los autosomas, lo cual podría indicar que en la población había menos machos que hembras. Esto porque, en las aves, las hembras son el sexo heterogamético (cromosomas sexuales ZZ en machos y ZW en hembras; Ellegren y Fridolfsson, 1997). Esta desviación en la proporción sexual puede haber influido en la diversidad genética de la especie, así como en la estabilidad poblacional de la misma (Eberhart-Phillips et al., 2017). Por último, ambos métodos sugieren que no existe estructura genética en *P. californica*. Es importante mencionar que, si bien los resultados obtenidos a partir de RAD-Seq sugieren que *P. melanura* y *P. californica* pertenecen a un mismo clúster genético, ello no es resultado de una mala identificación de los especímenes muestreados. El individuo de la especie hermana, la perlita del desierto, fue colectado fuera del área de distribución natural de la perlita de California. Así,

la falta de divergencia genética entre ambas especies en esta prueba pudo deberse a un error durante la preparación de la genoteca o el proceso de secuenciación.

Si bien este estudio nos brinda una idea más clara sobre el estado de las poblaciones de *P. california* y el impacto que han tenido los cambios climáticos globales en la demografía de las especies de zonas áridas del noroeste de América, reconocemos las limitaciones de este estudio. Por lo tanto, creemos que un muestreo exhaustivo que incluya tanto un mayor número de individuos como una mayor cantidad de localidades podría resolvernos dudas sobre la estructura genética de la especie. Asimismo, un genoma de mayor cobertura daría mayor resolución a los análisis de demografía histórica al incluir un mayor número de loci. Sin embargo, los resultados de este trabajo muestran que la distribución geográfica y los patrones genéticos de la perlita de California están asociados a los cambios climáticos y ambientales que tuvieron lugar en el periodo Cuaternario. Más aún, el hecho de que los patrones demográficos observados para *P. californica* se asociaran a los cambios del UMG y el UIG es alarmante bajo un escenario de cambio climático como el que vivimos actualmente.

Aún desconocemos si la desertificación global que enfrentamos hoy en día presentará un escenario favorable para aquellas aves con preferencias por ambientes áridos, pero los patrones y la escala temporal que mostró la perlita de California deben ser considerados como una señal de alerta para tomar urgentemente medidas de conservación mejor informadas para ésta y otras especies con nichos ecológicos similares. Una de las principales amenazas para la especie son los recurrentes incendios en la zona de la península, por lo que crear planes de control que mantengan la cantidad y magnitud de los incendios en temporada de sequía al mínimo podría beneficiar a la especie al reducir el posible aislamiento entre poblaciones debido a la falta de matorrales.

Referencias bibliográficas

- Aguirre-Liguori JA, Ramírez-Barahona S, Gaut BS. 2021. The evolutionary genomics of species' responses to climate change. Nat. Ecol. Evol. 5(10):1350–1360.
- Álvarez-Castañeda ST, Murphy RW. 2014. The endemic insular and peninsular species *Chaetodipus spinatus* (mammalia, heteromyidae) breaks patterns for Baja California. PLoS ONE 9(12):1–26.
- Bennett KD. 1990. Milankovitch Cycles and Their Effects on Species in Ecological and Evolutionary Time. Paleobiology 16(1):11–21.
- Brown JW. 1987. The Peninsular Effect in Baja California: An Entomological Assessment. J. Biogeogr. 14(4):359.
- Brown SM, Harrisson KA, Clarke RH, Bennett AF, Sunnucks P. 2013. Limited Population Structure, Genetic Drift and Bottlenecks Characterise an Endangered Bird Species in a Dynamic, Fire-Prone Ecosystem. PLoS ONE 8(4):e59732.
- Burg TM, Gaston AJ, Winker K, Friesen VL. 2006. Effects of Pleistocene glaciations on population structure of North American chestnut-backed chickadees. Mol. Ecol. 15(9):2409-2419.
- Drovetski Sv et al. 2018. A test of the European Pleistocene refugial paradigm, using a Western Palaearctic endemic bird species. Proc. R. Soc. B: Biol. Sci. 285(1889):20181606.
- Eberhart-Phillips LJ et al. 2017. Sex-specific early survival drives adult sex ratio bias in snowy plovers and impacts mating system and population growth. Proc. Natl. Acad. Sci. U.S.A. 114(27):E5474–E5481.
- Ellegren H, Fridolfsson A-K. 1997. Male–driven evolution of DNA sequences in birds. Nat. Genet. 17(2):182–184.
- Feng X, Park DS, Liang Y, Pandey R, Papeş M. 2019. Collinearity in ecological niche modeling: Confusions and challenges. Ecol. Evol. 9(18):10365–10376.

- Forsdick NJ et al. 2017. Genetic diversity and population differentiation within and between island populations of two sympatric Petroica robins, the Chatham Island black robin and tomtit. Conserv. Genet. 18:275–285.
- Graham MR, Santibáñez-López CE, Derkarabetian S, Hendrixson BE. 2020. Pleistocene persistence and expansion in tarantulas on the Colorado Plateau and the effects of missing data on phylogeographical inferences from RADseq. Mol. Ecol. 29(19):3684–3701.
- Herzschuh U. 2020. Legacy of the Last Glacial on the present-day distribution of deciduous versus evergreen boreal forests. Glob. Ecol. Biogeogr. 29(2):198–206.

Hewitt GM. 2000. The genetic legacy of the Quaternary ice ages. Nature 405(6789):907-913.

- Hewitt GM. 2001. Speciation, hybrid zones and phylogeography or seeing genes in space and time. Mol. Ecol. 10(3):537–549.
- Hewitt GM. 2004. Genetic consequences of climatic oscillations in the Quaternary. Philos. Trans.R. Soc. B: Biol. Sci. 359(1442):183–195.
- Hoskin CJ, Higgie M, McDonald KR, Moritz C. 2005. Reinforcement drives rapid allopatric speciation. Nature 437(7063):1353–1356.
- Jackson HA et al. 2022. Genomic erosion in a demographically recovered bird species during conservation rescue. Conserv. Biol. 36(4):e13918.
- Keller LF, Waller DM. 2002. Inbreeding effects in wild populations. Trends Ecol. Evol. 17(5):230– 241.
- Klimova A, Hoffman JI, Gutierrez-Rivera JN, Leon de la Luz J, Ortega-Rubio A. 2017. Molecular genetic analysis of two native desert palm genera, *Washingtonia* and *Brahea*, from the Baja California Peninsula and Guadalupe Island. Ecol. Evol. 7(13):4919–4935.
- Klinga P et al. 2019. Considering landscape connectivity and gene flow in the Anthropocene using complementary landscape genetics and habitat modelling approaches. Landsc. Ecol. 34(3):521–536.

- Leaché AD, Crews SC, Hickerson MJ. 2007. Two waves of diversification in mammals and reptiles of Baja California revealed by hierarchical Bayesian analysis. Biol. Let. 3(6):646–650.
- Mantooth SJ, Hafner DJ, Bryson RW, Riddle BR. 2013. Phylogeographic diversification of antelope squirrels (*Ammospermophilus*) across North American deserts. Biol. J. Linn. Soc. 109(4):949– 967.
- McCormack JE, Maley JM. 2015. Interpreting negative results with taxonomic and conservation implications: Another look at the distinctness of coastal California Gnatcatchers. Auk *132*(2):380–388.
- McDonough MM, Ferguson AW, Dowler RC, Gompper ME, Maldonado JE. 2022. Phylogenomic systematics of the spotted skunks (Carnivora, Mephitidae, Spilogale): additional species diversity and Pleistocene climate change as a major driver of diversification. Mol. Phylogenet. Evol. 167: 107266.
- Menezes L, Batalha-Filho H, Garda AA, Felgueiras Napoli M. 2021. Tiny treefrogs in the Pleistocene: Phylogeography of *Dendropsophus oliveirai* in the Atlantic Forest and associated enclaves in northeastern Brazil. J. Zoolog. Syst. Evol. Res. 59(1):179-194.

Miller J. 2010. Species distribution modeling. Geogr. Compass 4(6):490–509.

- Moncrieff GR, Hickler T, Higgins SI. 2015. Intercontinental divergence in the climate envelope of major plant biomes. Glob. Ecol. Biogeogr. 24(3):324–334.
- Morrone JJ. 2021. Biogeographic regionalisation of the Baja California biogeographic province, Mexico: A review. J. Nat. Hist. 55(5-6):365–379.
- Nason JD, Hamrick JL, Fleming TH. 2002. Historical vicariance and postglacial colonization effects on the evolution of genetic structure in *Lophocereus*, a Sonoran desert columnar cactus. Evolution 56(11):2214–2226.
- National Research Council. 1995. *Science and the Endangered Species Act*. Washington, DC: The National Academies Press.

- Ony M et al. 2021. Genetic diversity in North American *Cercis canadensis* reveals an ancient population bottleneck that originated after the last glacial maximum. Sci. Rep. 11:21803.
- Owens HL et al. 2013. Constraints on interpretation of ecological niche models by limited environmental ranges on calibration areas. Ecol. Modell. 263:10–18.
- Palstra FP, Ruzzante DE. 2008. Genetic estimates of contemporary effective population size: what can they tell us about the importance of genetic stochasticity for wild population persistence? Mol. Ecol. 17(15):3428–3447.
- Prates I et al. 2016. Inferring responses to climate dynamics from historical demography in neotropical forest lizards. Proc. Natl. Acad. Sci. U.S.A. 113(29):7978–7985.
- Pyron RA, Burbrink FT. 2010. Hard and soft allopatry: Physically and ecologically mediated modes of geographic speciation. J. Biogeogr. 37(10):2005–2015.
- Riddle BR, Hafner DJ, Alexander LF, Jaeger JR. 2000. Cryptic vicariance in the historical assembly of a Baja California Peninsular Desert biota. Proc. Natl. Acad. Sci. U.S.A. 97(26):14438– 14443.
- Riddle BR, Hafner DJ. 2006. Biogeografía histórica de los desiertos cálidos de norteamérica. In:
 Vázquez-Domínguez E, Hafner DJ, editors. *Genética y mamíferos mexicanos: presente y futuro: Bulletin 32*, 32. New Mexico: New Mexico Museum of Natural History and Science Bulletin. pp. 57–65.
- Sánchez-del Pino, I et al. (2020). High phylogeographic and genetic diversity of *Tidestromia lanuginosa* supports full-glacial refugia for arid-adapted plants in southern and central Coahuila, Mexico. Am. J. Bot. 107(9):1296–1308.
- Schiffels S, Wang K. 2020. MSMC and MSMC2: The Multiple Sequentially Markovian Coalescent.
 In: Dutheil JY, editor. *Statistical Population Genomics. Methods in Molecular Biology*, 2090.
 New York: Humana. pp. 147–166.

- Sedlock RL. 2003. Geology and tectonics of the Baja California Peninsula and adjacent areas. In: Johnson SE et al, editors. *Special Paper 374: Tectonic evolution of northwestern Mexico and the Southwestern USA*, 374. Colorado: Geological Society of America. pp. 1–42.
- Stewart JR, Lister AM, Barnes I, Dalen L. 2009. Refugia revisited: individualistic responses of species in space and time. Proc. R. Soc. B: Biol. Sci. 277(1682):661–671.
- Svenning J-C, Fløjgaard C, Marske KA, Nógues-Bravo D, Normand S. 2011. Applications of species distribution modeling to paleobiology. Quat. Sci. Rev. 30(21–22):2930–2947.
- Vázquez-Miranda H, Zink RM, Pinto BJ. 2022. Comparative phylogenomic patterns in the Baja California avifauna, their conservation implications, and the stages in lineage divergence. Mol. Phylogenet. Evol. 171:107466.
- Waltari E, Hijmans RJ, Peterson AT, Nyári AS, Perkins SL. 2007. Locating Pleistocene Refugia: Comparing Phylogeographic and Ecological Niche Model Predictions. PLoS ONE 2(7):563.
- White TA, Searle JB. 2007. Genetic diversity and population size: Island populations of the common shrew, *Sorex araneus*. Mol. Ecol. 16(10):2005–2016.
- Williams JW, Shuman BN, Webb T, Bartlein PJ, Leduc PL. 2004. Late-Quaternary vegetation dynamics in North America: Scaling from taxa to biomes. Ecol. Monogr. 74(2):309–334.
- Wilson JS, Pitts JP. 2012. Identifying Pleistocene refugia in North American cold deserts using phylogeographic analyses and ecological niche modelling. Diver. Distrib. 18(11):1139–1152.
- Hu Y et al. 2020. Genomic evidence for two phylogenetic species and long-term population bottlenecks in red pandas. Sci. Adv. 6(9): eaax5751.
- Zink RM, Barrowclough GF, Atwood JL, Blackwell-Rago RC. 2000. Genetics, taxonomy, and conservation of the threatened California Gnatcatcher. Conserv. Biol. 14(5):1394–1405.
- Zink RM, Groth JG, Vazquez-Miranda H, Barrowclough GF. 2013. Phylogeography of the California Gnatcatcher (*Polioptila californica*) using multilocus DNA sequences and ecological niche modeling: Implications for conservation. Auk 130(3):449–458.

Zink RM, Groth JG, Vázquez-Miranda H, Barrowclough GF. 2016. Geographic variation, null hypotheses, and subspecies limits in the California Gnatcatcher: A response to McCormack and Maley. Auk 133(1):59–68.

Apéndices

Supplementary Material

TITLE

Synchronous demographic and distribution shifts through time in an endangered desert specialist

AUTHORS AND AFFILIATIONS

Camila Martínez-Avila^{1,2}, Carolina Granados Mendoza³, Ella Vázquez Domínguez⁴, and Hernán Vázquez-Miranda^{1,*}.

¹Colección Nacional de Aves, Departamento de Zoología, Instituto de Biología, Universidad Nacional Autónoma de México, CDMX 04510.

²Posgrado en Ciencias Biológicas, Universidad Nacional Autónoma de México, CDMX 04510.

³Departamento de Botánica, Instituto de Biología, Universidad Nacional Autónoma de México,

CDMX 04510.

⁴Laboratorio de Genética y Ecología, Departamento de Ecología de la Biodiversidad, Instituto de Ecología, Universidad Nacional Autónoma de México, 04510.

*Corresponding author: E-mail: hernan@ib.unam.mx.

Supplementary Tables

Sample ID	Location	GBS	RAD-Seq	Voucher location
JK99-023	Riverside,	X	X	Bell Museum, University of Minnesota
	USA			
JK99-024	Riverside, California, USA	Х		Bell Museum, University of Minnesota
JK99-025	Riverside, California, USA	Х	Х	Bell Museum, University of Minnesota
BCHVM-064	Cataviña, Baja California, México		X	Museo de Zoología, Facultad de Ciencias, UNAM, México
BCHVM-070	Cataviña, Baja California, México	X		Museo de Zoología, Facultad de Ciencias, UNAM, México
BCHVM-085	Cataviña, Baja California, México	Х	Х	Museo de Zoología, Facultad de Ciencias, UNAM, México
BCHVM-087	Cataviña, Baja California, México	Х		Museo de Zoología, Facultad de Ciencias, UNAM, México
BCHVM-089	Cataviña, Baja California, México	Х		Museo de Zoología, Facultad de Ciencias, UNAM, México
BCHVM-118	Cataviña, Baja California, México	Х		Museo de Zoología, Facultad de Ciencias, UNAM, México
BCHVM-135	Santa Rosalillita, Baja California,	X		Museo de Zoología, Facultad de Ciencias, UNAM, México

Table S1. Sampled individuals, locations and sequencing method

	México			
BCHVM-136	Santa Rosalillita, Baja California, México	X	Х	Museo de Zoología, Facultad de Ciencias, UNAM, México
BCHVM-137	Santa Rosalillita, Baja California, México	X		Museo de Zoología, Facultad de Ciencias, UNAM, México
BCHVM-138	Santa Rosalillita, Baja California, México	X	X	Museo de Zoología, Facultad de Ciencias, UNAM, México
BCHVM-139	Santa Rosalillita, Baja California, México	X	X	Museo de Zoología, Facultad de Ciencias, UNAM, México
BCHVM-199	El Cien, Baja California Sur, México	Х	X	Museo de Zoología, Facultad de Ciencias, UNAM, México
BCHVM-201	El Cien, Baja California Sur, México	Х	Х	Museo de Zoología, Facultad de Ciencias, UNAM, México
BCHVM-203	El Cien, Baja California Sur, México	X	X	Museo de Zoología, Facultad de Ciencias, UNAM, México
BCHVM-204	El Cien, Baja California Sur, México	X	X	Museo de Zoología, Facultad de Ciencias, UNAM, México
BCHVM-213	El Cien, Baja California Sur,	Х		Museo de Zoología, Facultad de Ciencias, UNAM, México

	México			
BCHVM-214	El Cien, Baja California Sur, México	X		Museo de Zoología, Facultad de Ciencias, UNAM, México
BCHVM-222	Cabo Pulmo, Baja California Sur, México	X	X	Museo de Zoología, Facultad de Ciencias, UNAM, México
BCHVM-223	Cabo Pulmo, Baja California Sur, México	X	X	Museo de Zoología, Facultad de Ciencias, UNAM, México
BCHVM-224	Cabo Pulmo, Baja California Sur, México	X	X	Museo de Zoología, Facultad de Ciencias, UNAM, México
RMZ5259	Sonora, México	X	Х	Bell Museum, University of Minnesota

Supplementary Figures



Fig. S1. Sampled localities of the California gnatcatcher. The individuals sequenced in this study came from the five localities indicated by the red dots. From North to South the locations are: Riverside, California, EE.UU.; Cataviña, Baja California; Santa Rosalillita, Baja California; El Cien, Baja California Sur; and Cabo Pulmo, Baja California Sur.



Fig. S2. Cross-validation error for the different number of evaluated clusters (k). For the GBS data we evaluated k values from 1 to 6 on *ADMIXTURE*. The red dotted line indicates that k=2 is the number of clusters that has the highest probability of explaining the data.



Fig. S3. Population structure for *P. californica* **based on RAD-Seq data.** The figures depict to which genetic group the sampled individuals belong showing that all the smaples grouped in a single cluster. A and B illustrate the results obtained with *ADMIXTURE* and *fastSTRUCTURE*, respectively. While C shows the genetic coordinates of the samples based on a DAPC analysis. LD: linear discriminant.



Fig. S4. Historical demography of *P. californica*'s autosomes based on RAD-Seq data with 20 free parameters. The red line shows the demographic trend of the California gnatcatcher sampled locations. The light blue rectangle indicates the extent of the Last Glacial Period, while the dark blue line represents the duration of the Last Glacial Maximum.



Fig. S5. Historical demography of *P. californica*'s autosomes based on RAD-Seq data with 28 free parameters. The red line shows the demographic trend of the California gnatcatcher sampled locations. The light blue rectangle indicates the extent of the Last Glacial Period, while the dark blue line represents the duration of the Last Glacial Maximum.



Fig. S6. Historical demography of *P. californica*'s autosomes based on GBS data with 18 free parameters. The red dark line shows the demographic trend of the California gnatcatcher sampled locations and the light red lines are bootstrap replicates of the original run. The light blue rectangle indicates the extent of the Last Glacial Period, while the dark blue line represents the duration of the Last Glacial Maximum.



Fig. S7. Historical demography of *P. californica* **for the sex Z chromosome based on RAD-Seq data with 6 free parameters.** The red line shows the demographic trend of the California gnatcatcher sampled locations. The light blue rectangle indicates the extent of the Last Glacial Period, while the dark blue line represents the duration of the Last Glacial Maximum.



Fig. S8. Historical demography of *P. californica* **for the sex Z chromosome based on RAD-Seq data with 7 free parameters.** The red line shows the demographic trend of the California gnatcatcher sampled locations. The light blue rectangle indicates the extent of the Last Glacial Period, while the dark blue line represents the duration of the Last Glacial Maximum.



Fig. S9. Historical demography of *P. californica* **for the sex Z chromosome based on GBS data with 6 free parameters.** The red line shows the demographic trend of the California gnatcatcher sampled locations and the light red lines are bootstrap replicates of the original run. The light blue rectangle indicates the extent of the Last Glacial Period, while the dark blue line represents the duration of the Last Glacial Maximum.



Fig. S10. ROC curve for the *MaxEnt* **model based on the current bioclimatic variables.** The red line depicts the cumulative increase between the specificity (false positive rate) in the *x* axis and the sensibility (true positive rate) in the *y* axis. The Area Under the Curve (AUC) value was 0.956.